DESIGN, ANALYSIS AND EXPERIMENTAL VERIFICATION OF CENTER-POINT-CLAMPED AC-AC CONVERTER

by

Pankaj Kumar Bhowmik

A thesis submitted to the faculty of The University of North Carolina at Charlotte in partial fulfillment of the requirements for the degree of Master of Science in Electrical Engineering

Charlotte

2015

Approved by:

Dr. Madhav Manjrekar

Dr. Robert Cox

Dr. Badrul Chowdhury

©2015 Pankaj Kumar Bhowmik ALL RIGHTS RESERVED

ABSTRACT

PANKAJ KUMAR BHOWMIK. Design, analysis and experimental verification of Center Point Clamped AC-AC Converter. (Under the direction of Dr. MADHAV MANJREKAR)

With the increasing use of electronic loads by the consumers, the utilities are facing more and more power quality issues. The advancement in semiconductor industry has led to introduction of power electronic devices as solutions to the power quality issues. But the implementation of these devices at utility scale voltage still remains challenging due to lack of power semiconductor devices rated at utility scale voltages. This thesis proposes a novel topology of power converter that uses semiconductor devices rated at about half the magnitude of utility scale voltage. This novel converter topology would help realizing a power electronic converter for utility scale voltages with easily available semiconductor devices that are rated at half the utility scale voltages.

This thesis investigates a center point clamped ac-ac topology for direct ac-ac power conversion. This converter topology introduces a unique methodology to clamp the voltage to the mid-point of the input transformer thus reducing the voltage blocking requirements on the bi-directional switches employed in such direct ac-ac converter. This thesis also discusses the considerations that are taken into account while designing an input filter for Center-Point-Clamped AC-AC Converter. The switched mode regulators due to their switching action feed in unwanted harmonic currents to the input side. These switching harmonics are attenuated by the designed input filter. But the addition of an input filter changes the transfer functions associated with the converter, which may degrade the performance of the converter and may even destabilize the system. The dynamic analysis of the Center-Point-Clamped AC-AC Converter with an input filter has

been discussed in the thesis. A feedback controller has been designed for closed loop control of the converter. Simulation results presented in the paper verify that the proposed converter offers high power transfer efficiency compared to conventional ac-dc-ac converters, and smaller output voltage ripple along with reduced voltage stress on the bidirectional switches when compared to other direct ac-ac converters. Also, it is seen that the designed input filter reduces the harmonic distortion of the input current, maintains the stability of the system as well as does not degrade the performance of the Center-Point-Clamped AC-AC converter. Experimental results demonstrating the Center-Point-Clamped mechanism of the converter as well as closed loop control of the AC-AC Converter are also presented in the paper.

ACKNOWLEDGEMENTS

I would like to express my gratitude towards my parents, my sister for believing in me and making it possible for me to follow my dream of pursuing higher education.

I would like to thank my advisor, Dr. Madhav Manjrekar, for his guidance and encouragement, without which, this thesis would not have been possible. His words can always inspire me and take me to higher level of thinking in my future endeavors.

It would be my duty to extend my gratitude to the committee members Dr. Robert Cox and Dr. Badrul Chowdhury for taking time to be on my committee and assess my work.

I am thankful to EPIC (Energy Production and Infrastructure Center) for its generous support.

Finally I am thankful to my lab mates, my friends and well-wishers who have always been there for me throughout the process.

TABLE OF CONTENTS

LIST OF FIGURES	viii
LIST OF ABBREVIATIONS	xiv
CHAPTER 1 : INTRODUCTION	1
1.1. Introduction	1
1.2. Power Quality Issues in the Modern Utility Grid	1
1.3. Comparison of Direct and Indirect AC-AC Converters	2
1.4. Challenges and Thesis Motivation	3
1.5. Organization of Thesis	4
CHAPTER 2 : REVIEW OF AC-AC DIRECT POWER CONVERTERS	8
2.1. Introduction	8
2.2. Single Phase AC Chopper Circuit with Shunt Connected Bidirectional Switches	8
2.3. AC Chopper Circuit with Shunt Connected Bidirectional Switches	9
2.4. Classical PWM Three Phase AC Chopper Circuit	12
2.5. Vector Switching Converter	13
2.6. Matrix Converter	14
2.7. Phase and Amplitude Controllable Transformer	16
2.8. Multilevel Direct AC Converter	17
CHAPTER 3 : CENTER-POINT-CLAMPED AC-AC BUCK CONVERTER	20
3.1. Introduction	20
3.2. Evolution of Center-Point-Clamped AC-AC Converter	21
3.3. Operating Principle of Center-Point-Clamped AC-AC Buck Converter	23
3.4. Methodology for Balancing Voltages across Input capacitors	25

CHAPTER 4 : CIRCUIT DESIGN OF CENTER-POINT-CLAMPED AC-AC BUCK CONVERTER	26
4.1. Introduction	26
4.2. Circuit Description	26
4.3. Power Circuit Design	27
CHAPTER 5 : DYNAMIC ANALYSIS OF CENTER-POINT-CLAMPED AC-AC BUCK CONVERTER	37
5.1. Introduction	37
5.2. Circuit Description	37
5.3. State Space Equations	38
5.4. Dynamic Analysis	39
CHAPTER 6 : CONTROLLER DESIGN FOR CLOSED LOOP CONTROL OF CENTER-POINT-CLAMPED AC-AC CONVERTER	60
6.1. Introduction	60
6.2. Circuit Description	60
6.3. Compensator Design	61
CHAPTER 7 : INPUT FILTER CONSIDERATIONS FOR CENTER-POINT- CLAMPED AC-AC CONVERTER	64
7.1. Introduction	64
7.2. Modified Center-Point-Clamped AC-AC Converter Circuit	65
7.3. Description of Modified Center-Point-Clamped AC-AC Buck Converter Circuit	66
7.4. Dynamic Analysis of Modified Center-Point-Clamped AC-AC Buck Converter	66
7.5. Input Filter Considerations for Center-Point-Clamped AC-AC Converter Circuit	77

vii

	viii
CHAPTER 8 : SIMULATION RESULTS	85
8.1. Introduction	85
8.2. Matlab-Simulink Model of Center-Point-Clamped AC-AC Converter	86
8.3. Operational Simulation Results	88
8.4. Open Loop Simulation Results	92
8.5. Closed Loop Simulation Results	96
CHAPTER 9 : EXPERIMENTAL RESULTS	103
9.1. Introduction	103
9.2. Experimental Set-up Description	104
9.3. Operational Experimental Results	110
9.4. Open Loop Experimental Results	114
9.5. Closed Loop Experimental Results	119
CHAPTER 10 : FAMILY OF CENTER-POINT-CLAMPED AC-AC CONVERTERS	124
10.1. Introduction	124
10.2. Center-Point-Clamped Direct AC-AC Buck Converter	125
10.3. Center-Point-Clamped Direct AC-AC Boost Converter	128
10.4. Center-Point-Clamped Direct AC-AC Buck-Boost Converter	131
10.5. Center-Point-Clamped Direct AC-AC Ćuk Converter	134
CHAPTER 11 : CONCLUSIONS	138
REFERENCES	140

LIST OF FIGURES

FIGURE 2	2.1: Simplified circuit schematic of the single phase of original ac chopper circuit with shunt connected bidirectional switch.	8
FIGURE 2	2.2: Simplified circuit schematic of the single phase of full ac chopper circuit with series and shunt connected bidirectional switches.	10
FIGURE 2	2.3: Simplified circuit schematic of three-phase AC chopper with six bidirectional switches realized by twelve bipolar junction transistors.	11
FIGURE 2	2.4: Simplified circuit schematic of three-phase AC chopper circuit realized with six insulated gate bipolar transistors.	12
FIGURE 2	2.5: Simplified circuit schematic of Vector Switching Converter (VeSC).	13
FIGURE 2	2.6: Simplified circuit schematic of a three phase to three phase matrix converter	15
FIGURE 2	2.7: Simplified circuit schematic of a Phase Amplitude Controllable Transformer	17
FIGURE 2	2.8: Simplified circuit schematic of multilevel ac-ac converter.	18
FIGURE 3	1: Simplified circuit schematics showing evolution of Center-Point- Clamped AC-AC Converter.	21
FIGURE 3	2.2: Simplified circuit schematic of an ac-ac converter.	21
FIGURE 3	3.3: Simplified circuit schematic of single phase center point clamped ac-ac converter.	22
FIGURE 3	3.4: Simplified circuit schematic of single phase center point clamped ac-ac converter.	24
FIGURE 4	.1: Simplified circuit schematic of a single phase of center-point- clamped ac-ac converter.	26
FIGURE 4	.2: Equivalent circuit of center-point-clamped ac-ac converter in interval $T_{1.}$	28
FIGURE 4	.3: Equivalent circuit of center-point-clamped ac-ac converter in interval T ₂ .	29
FIGURE 4	.4: Equivalent circuit of center-point-clamped ac-ac converter in interval T_1 .	32

FIGURE 4.5: Equivalent circuit of center-point-clamped ac-ac converter in interval T_2 .	x 33
FIGURE 5.1: Simplified circuit schematic of a single phase of center-point- clamped ac-ac converter.	37
FIGURE 5.2: Equivalent circuits of a center-point-clamped ac-ac converter in interval T_1 .	39
FIGURE 5.3: Equivalent circuit of a center-point-clamped ac-ac converter in interval T_2 .	41
FIGURE 5.4: Magnitude and phase plots of transfer function, G_d for output voltage range 0 to $v_{in}/2$.	48
FIGURE 5.5: Magnitude and phase plots of transfer function, G_v for output voltage range 0 to $v_{in}/2$.	49
FIGURE 5.6: Equivalent circuit of a center-point-clamped ac-ac converter in interval T_1 .	49
FIGURE 5.7: Equivalent circuits of a center-point-clamped ac-ac converter in interval T ₂ .	51
FIGURE 5.8: Magnitude and phase plots of transfer function, G_d for output voltage range $v_{in}/2$ to v_{in} .	58
FIGURE 5.9: Magnitude and phase plots of transfer function, G_v for output voltage range $v_{in}/2$ to v_{in} .	59
FIGURE 6.1: Simplified circuit schematic of a single phase of center-point- clamped ac-ac converter.	60
FIGURE 6.2: Step response plot of uncompensated converter output voltage to duty cycle transfer function, G_d .	62
FIGURE 6.3: Control block diagram of center-point-clamped ac-ac converter.	62
FIGURE 6.4: Magnitude and phase plots of transfer function, $G_d * G_c$ for output voltage range 0 to v_{in} .	63
FIGURE 6.5: Step response plot of transfer function, $G_d * G_c$ for output voltage range 0 to v_{in} .	63
FIGURE 7.1: Simplified circuit schematic of Center-Point-Clamped AC-AC converter with input filter.	65

FIGURE 7.2: Simplified circuit schematic of Center-Point-Clamped AC-AC converter with input filter.	xi 77
FIGURE 7.3: Generalized block schematic of a converter with an input filter in addition to the output filter	79
FIGURE 7.4: Bode plots for $Z_o(s)$, $Z_N(s)$ and $Z_D(s)$	81
FIGURE 7.5: Bode plot for $G_{vd}(s)$	82
FIGURE 7.6: Bode plot of uncompensated transfer function, $G'_{vd}(s)$	82
FIGURE 7.7: Bode plot of compensated transfer function $G'_{vd}(s) * G_c(s)$	83
FIGURE 8.1: Simplified circuit schematic of Center-Point-Clamped AC-AC converter with input filter.	85
FIGURE 8.2: Matlab-Simulink model of Center-Point-Clamped AC-AC converter with input filter.	87
FIGURE 8.3: Unfiltered Output Voltage waveforms when $v_{oref} = 0.3 v_{in}$	88
FIGURE 8.4: Unfiltered Output Voltage waveforms when $v_{oref} = 0.8v_{in}$	89
FIGURE 8.5: Filtered Output Voltage waveforms when $v_{oref} = 0.8v_{in}$	89
FIGURE 8.6: Filtered Output Voltage waveforms when $v_{oref} = 0.8v_{in}$	90
FIGURE 8.7: Voltages across S2 and S3	91
FIGURE 8.8: Voltages across S_1 and S_4	91
FIGURE 8.9: Filtered output voltage waveform for D=0.25	92
FIGURE 8.10: Filtered output current waveform for D=0.25	92
FIGURE 8.11: Input source voltage waveform for D=0.25	93
FIGURE 8.12: Waveform of input current drawn from the source for D=0.25	93
FIGURE 8.13: Filtered output voltage waveform for D=0.75	94
FIGURE 8.14: Filtered output current waveform for D=0.75	94
FIGURE 8.15: Input source voltage waveform for D=0.75	95

	xii
FIGURE 8.16: Waveform of input current drawn from the source for D=0.75	95
FIGURE 8.17: Filtered output voltage waveform for $v_{oref} = 1000V$	96
FIGURE 8.18: Filtered output voltage waveform for $v_{oref} = 3000$ V	96
FIGURE 8.19: Change of operating point within Region 1 (1000V to 1500V)	97
FIGURE 8.20: Change of operating point within Region 2 (2500V to 3000V)	98
FIGURE 8.21: Change of operating point from Region 1 (1500V) to Region 2 (3000V)	98
FIGURE 8.22: Line regulation: performance during 10% increase in line voltage	99
FIGURE 8.23: Line regulation: performance during 10% decrease in line voltage	100
FIGURE 8.24: Load regulation: performance during 50% decrease in load current	101
FIGURE 8.25: Load regulation: performance during 50% decrease in load current	101
FIGURE 9.1: Picture of experimental prototype in the laboratory	104
FIGURE 9.2: Picture of LabVIEW VI Block Diagram window implementing the gate driver logic	105
FIGURE 9.3: Driver Logic Signal waveform for switches, S_1 , S_2 , S_3 , and S_4 when $v_{oref}=0.3v_{in}$	106
FIGURE 9.4: Driver Logic Signal waveform for switches, S_1 , S_2 , S_3 , and S_4 when $v_{oref}=0.8v_{in}$	107
FIGURE 9.5: User Interface for entering reference output voltage, v_{oref} command input.	107
FIGURE 9.6: Picture of the LabVIEW VI sensing the reference output voltage, v_{oref} from the slider input	108
FIGURE 9.7: Picture of the LabVIEW VI sensing output voltage, v_o using Phase Locked Loop	109
FIGURE 9.8: Picture of the LabVIEW VI implementing the feedback controller	110
FIGURE 9.9: Unfiltered Output Voltage waveforms when $v_{oref} = 0.3 v_{in}$	111

FIGURE 9.10: Unfiltered Output Voltage waveforms when $v_{oref} = 0.8v_{in}$	xiii 111
FIGURE 9.11: Filtered Output Voltage waveforms when $v_{oref} = 0.3v_{in}$	112
FIGURE 9.12: Filtered Output Voltage waveforms when $v_{oref} = 0.8v_{in}$	112
FIGURE 9.13: Voltages across S2 and S3	113
FIGURE 9.14: Voltages across S_1 and S_4	114
FIGURE 9.15: Filtered output voltage waveform for D=0.25	115
FIGURE 9.16: Filtered output current waveform for D=0.25	115
FIGURE 9.17: Input source voltage waveform for D=0.25	116
FIGURE 9.18: Waveform of input current drawn from the source for D=0.25	116
FIGURE 9.19: Filtered output voltage waveform for D=0.75	117
FIGURE 9.20: Filtered output current waveform for D=0.75	117
FIGURE 9.21: Input source voltage waveform for D=0.75	118
FIGURE 9.22: Waveform of input current drawn from the source for D=0.75	118
FIGURE 9.23: Change of operating point within Region 1 (20V to 50V).	119
FIGURE 9.24: Change of operating point within Region 2 (70V to 90V).	120
FIGURE 9.25: Change of operating point from Region 1 (20V) to Region 2 (90V).	120
FIGURE 9.26: Line regulation: performance during step increase in line voltage when $v_{oref} = 80$ V.	121
FIGURE 9.27: Line regulation: performance during step decrease in line voltage when $v_{oref} = 80$ V.	121
FIGURE 9.28: Load regulation: performance during step decrease in load current when $v_{oref} = 80$ V.	122
FIGURE 9.29: Load regulation: performance during step increase in load current when $v_{oref} = 80$ V.	123
FIGURE 10.1: Simplified Circuit Schematic of a Center-Point-Clamped Direct	125

AC-AC Buck Converter

FIGURE 10.2: Filtered output voltage waveform for D=0.25	126
FIGURE 10.3: Filtered output voltage waveform for D=0.75	127
FIGURE 10.4: Simplified Circuit Schematic of a Center-Point-Clamped Direct AC-AC Boost Converter	128
FIGURE 10.5: Filtered output voltage waveform for D=0.25	130
FIGURE 10.6: Filtered output voltage waveform for D=0.75	130
FIGURE 10.7: Simplified Circuit Schematic of a Center-Point-Clamped Direct AC-AC Buck-Boost Converter	131
FIGURE 10.8: Filtered output voltage waveform for D=0.25	133
FIGURE 10.9: Filtered output voltage waveform for D=0.75	133
FIGURE 10.10: Simplified Circuit Schematic of a Center-Point-Clamped Direct AC-AC Ćuk Converter	134
FIGURE 10.11: Filtered output voltage waveform for D=0.25	136
FIGURE 10.12: Filtered output voltage waveform for D=0.75	136

xiv

LIST OF ABBREVIATIONS

BJT	Bipolar Junction Transistor
IGBT	Integrated Gate Bipolar Transistor
SOA	Safe Operating Area
PWM	Pulse Width Modulation
VeSC	Vector Switching Converter
PACT	Phase Amplitude Controllable Transformer
TACC	Thin AC Converters
SISO	Single Input/Single Output
GUI	Graphical User Interface
EMI	Electro-Magnetic Interference
NI	National Instruments
GPIC	General Purpose Inverter Controller
VI	Virtual Instrument
QML	Query Mark-up Language
MSO	Mixed Signal Oscilloscope
FACTS	Flexible AC Transmission Systems

CHAPTER 1 : INTRODUCTION

1.1. Introduction

This chapter presents a brief introduction and organization of the thesis. Importance of power control equipment to improve power quality is emphasized in Section 1.2. Section 1.3 presents the importance of direct ac-ac converters. Section 1.4 discusses the challenges faced when operating at high voltage and high power levels as well as the primary motivation for the thesis. The concluding section presents organization of the thesis.

1.2. Power Quality Issues in the Modern Utility Grid

Increasing use of loads supplied by electronic power converters in industry has led to growing problems with reliability of the power supply. Computers, adjustable speed drives and automated manufacturing processes are very susceptible to voltage sags, spikes and power interruptions [1], [2]. The need for a reliable and high quality power supply has led to the increased adoption of electronic power conditioning devices and uninterruptible power supplies, which regulate voltages within acceptable tolerances [1], [15]-[17]. Another important issue affecting the modern utility grid is the addition of renewable energy sources to satisfy the surging energy demands. Penetration of renewable energy resources, while critical from the perspective of meeting increasing energy demands, adds to grid congestion and affects system operation [2]. Over the last few decades there has been a growing need for enhanced grid control capabilities

attributed to these issues. Advances in power semiconductor devices have made it possible for utilities to use a variety of power electronics based power control equipment to improve power quality. But the implementation of these power electronic solutions at utility scale voltages remain challenging as semiconductor devices at utility scale voltages are not readily available in the market. Also, these power electronic solutions are costly and less efficient, as they are quite complex. Thus, it may be noted that owing to high cost, complexity and poor reliability these solutions have seen minimal market penetration.

1.3. Comparison of Direct and Indirect AC-AC Converters

The direct ac-ac converter is one of the candidates in realizing a low cost power quality solution with high system reliability [1], [2], [10], [11]. This direct ac-ac conversion is unlike an inverter-based ac-dc-ac converter where additional energy storage is required (in the form of a dc link capacitor). In an indirect ac-ac converter, the power conversion occurs in several stages, like ac power is rectified to dc power and then dc power is inverted to ac power. The intermediate stages in an indirect ac-ac converter involve additional energy storage capacitors. Although such indirect ac-ac converters are simple to control and are cost competitive, they suffer from disadvantages such as requirement of storage element, viz. capacitor, which are bulky in size. Also, the multiple stages in an indirect ac-ac converter often involve a number of active and passive devices, which reduces its efficiency as well as life span. On the other hand, direct ac-ac power converters, without energy storage elements offer advantages such as, single-stage power conversion, superior harmonic content, better efficiency and smaller footprint [2], [10], [11]. Although a number of direct ac-ac converter topologies have been reported in

literature, yet the implementation of these converters at utility scale voltages and power have been a critical issue due to lack of availability of semiconductor devices rated at utility level voltages and power.

1.4. Challenges and Thesis Motivation

Operation of power conditioning systems such as direct ac-ac converters at high voltage and power levels requires semiconductor devices with high voltage blocking and high current carrying capabilities. Commercially available Insulated Gate Bipolar Transistors (IGBTs) rated up to 1700V/1200A that are low cost have been a workhorse for industrial applications operating at 480Vac [2]. These devices operate with relatively high switching frequencies. In order to operate direct ac-ac converter at higher voltage levels in the range of 4.16kV-13.8kV, it requires semiconductor devices rated at those higher voltage levels. As for instance, the phase-neutral peak voltage in a 4.16kV system is 3.4kV. To ensure that devices work reliably within their Safe Operating Area (SOA), it is required to employ either a 6.6kV IGBT or a series connection of two 3.3kV devices for this application. Because of lack of widely available 6.6kV devices, an alternative has been to employ series connection of 3.3kV IGBTs. This increases complexity and cost of system due to issues associated with static and dynamic voltage sharing of series connected IGBTs [2], [18]-[22]. It may be observed that as the voltage ratings of the semiconductor devices increase, the switch on resistance increases, thereby increasing the switching losses and reducing the efficiency of the semiconductor switch based power conditioning devices. An effective solution to all these challenges pertaining to construct a reliable and efficient power conditioning device has been proposed in this thesis in the form of a novel topology of direct ac-ac converter termed as Center-Point-Clamped AC-

AC Direct Power Converter. With the proposed center-point-clamped ac-ac converter topology, it has been shown that one can reliably construct an efficient power conditioning device for utility scale voltage and power ratings using semiconductor devices rated at half the utility scale voltages and power.

1.5. Organization of Thesis

This section presents an overview of the organization of thesis as follows:

Chapter 1: Introduction

This chapter starts by describing the importance of power control equipment to improve power quality in Section 1.2. The advantages of direct ac-ac converters over traditional ac-dc-ac converters are briefly discussed in Section 1.3. The challenges faced at higher voltage and power levels leading to the thesis motivation are discussed in Sections 1.4. The organization of the thesis is presented in Section 1.5.

Chapter 2: Review of AC-AC Direct Power Converters

This chapter presents the evolution of different types of AC-AC Direct Power Converters. A brief overview of the chapter is presented in Section 2.1. The single phase ac chopper circuit with shunt connected bidirectional switches is discussed in Section 2.2. The single phase and three phase ac chopper circuit with shunt and series connected bidirectional switches are studied in Section 2.3. The three phase AC chopper circuit constituting classical model of Pulse Width Modulation (PWM) based converter topologies, viz. buck, boost, and buck-boost converter are analyzed in Section 2.4. The Vector Switching Converter topology for M sources and N loads is discussed in Section 2.5. A generalized model of matrix converter is examined in Section 2.6. The Phase and Amplitude Controllable Transformer and a multilevel topology of AC-AC converter are studied in Section 2.7 and Section 2.8 respectively.

Chapter 3: Center-Point-Clamped AC-AC Buck Converter

This chapter introduces the novel topology of Center-Point-Clamped AC-AC Buck Converter. The functional capability of Center-Point-Clamped AC-AC Buck Converter is presented in Section 3.1. The working principle of Center-Point-Clamped AC-AC Buck Converter along with its switching sequence is discussed in Section 3.2. The principle of voltage balancing across the input capacitors is analyzed in Section 3.3.

Chapter 4: Circuit Design of Center-Point-Clamped AC-AC Buck Converter

This chapter discusses the circuit design principles followed to derive the various circuit parameters for the Center-Point-Clamped AC-AC Buck Converter. The preface to the chapter is presented in Section 4.1. The simplified circuit schematic of the converter is described in Section 4.2. The design of the converter circuit parameters is analyzed in Section 4.3.

Chapter 5: Dynamic Analysis of Center-Point-Clamped AC-AC Buck Converter

This chapter examines the dynamic analysis of the Center-Point-Clamped AC-AC Buck Converter. A brief overview of the chapter is presented in Section 5.1. The simplified circuit schematic of the converter is described in Section 5.2. The state space equations for the converter are obtained in Section 5.3. The dynamic analysis of the converter transfer functions is presented in Section 5.4.

Chapter 6: Controller Design for closed loop control of Center-Point-Clamped AC-AC Converter This chapter deals with the design of a feedback controller for closed loop operation of the Center-Point-Clamped AC-AC Buck Converter. The motivation behind this chapter is mentioned in Section 6.1. The simplified circuit schematic of the converter is shown in Section 6.2. The design of the feedback controller has been analyzed in Section 6.3.

Chapter 7: Input Filter considerations for Center-Point-Clamped AC-AC Converter

This chapter examines the considerations that have to be taken care of while designing an input filter for the converter so that the unregulated input side power supply does not get affected due to harmonics generated by the switching action of the converter. A brief preface to the chapter is presented in Section 7.1. The modified simplified circuit schematic of the converter is presented and described in Section 7.2 and Section 7.3. The dynamic analysis of the modified converter circuit is presented in Section 7.4. The considerations for designing the input filter for the converter are analyzed in Section 7.5.

Chapter 8: Simulation Results

This chapter presents the simulation results verifying the efficacy of the Center-Point-Clamped AC-AC Buck Converter. An introduction to the chapter is provided in Section 8.1. The simulation model of the converter is presented in Section 8.2. The waveforms supporting the operating principle of the converter are analyzed in Section 8.3. The open loop and closed loop simulation results are presented in Section 8.4 and Section 8.5 respectively.

Chapter 9: Experimental Results

This chapter discusses the experimental results obtained from a laboratory prototype verifying the practical realization and efficacy of the Center-Point-Clamped AC-AC

Buck Converter. The objective of this chapter is shown in Section 9.1. The experimental set-up is discussed in Section 9.2. The experimental waveforms explaining the operating principle of the converter are presented in Section 9.3. The open loop and closed loop experimental results are presented in Section 9.4 and Section 9.5.

Chapter 10: Family of Center-Point-Clamped AC-AC Converters

This chapter starts by stating the advantages of the family of Center-Point-Clamped AC-AC Direct Power Converter in Section 10.1. The operating principles and the switching methodologies of family of Center-Point-Clamped AC-AC Converters, viz. buck converter, boost converter, buck-boost converter, Ćuk converter are presented in Sections 10.2, 10.3, 10.4, and 10.5 respectively.

Chapter 11: Conclusions

This chapter contains a brief summary of the overall thesis. It also highlights the arguments and solutions presented in the thesis.

CHAPTER 2 : REVIEW OF AC-AC DIRECT POWER CONVERTERS

2.1. Introduction

A brief review of direct ac-ac converters has been presented in the following sections. Although a number of direct ac-ac converters have been reported in literature, selected topologies of converters that are more practical are discussed in the following sections.

2.2. Single Phase Ac Chopper Circuit with Shunt Connected Bidirectional Switches One of the first appearances of ac chopper circuit can be traced back to 1968 when it was employed for load voltage regulation applications [3]. In this paper, a technique has been introduced, based on forced ignition and extinction of thyristor based switches for ac voltage regulation. A schematic circuit diagram of the proposed technique is shown in figure 2.1.



Figure 2.1: Simplified circuit schematic of the single phase of original ac chopper circuit with shunt connected bidirectional switch.

A simplified circuit schematic of one of the earliest known ac chopper circuit has been shown in figure 2.1. It has a switch S which is series with the load. γ is the switching period of the switch S. The switch S is closed during the time interval, θ and opened during the rest of the switching time period, i.e. time interval $\gamma - \theta$. Thus, one cycle of the input voltage may be divided into integral multiples of the switching period. This technique is one of the fundamental PWM based ac voltage regulation applicable in resistive loads. However, it has inherent advantages in case of inductive loads and discontinuous load current as the circuit provides a means to discharge the magnetic energy stored in the inductor. During the half-cycle in which terminal A is at positive potential and S is switched off, thyristor, T_1 is switched on and T_2 is switched off. During the second half-cycle in which terminal B is at positive potential and S is switched on, thyristor, T_2 is switched on and T_1 is switched off. Thus, this methodology enables the prevention of exchange of reactive power between load and source owing to the discharge of magnetic energy near the load. The arrangement of thyristors T_1 and T_2 in back to back configuration acting as bidirectional switches is one of the earliest evidence of ac voltage regulation based on bidirectional switches that has been reported in the literature.

2.3. AC Chopper Circuit with Shunt Connected Bidirectional Switches

Extending the concept of ac voltage regulation further, a combination of shunt and series bidirectional switches realizing a full ac chopper circuit has been reported in [4] for power control applications.



Figure 2.2: Simplified circuit schematic of the single phase of full ac chopper circuit with series and shunt connected bidirectional switches.

The single phase version of ac chopper may be seen in figure 2.2. The bidirectional switches, S_1 and S_2 are closed and opened in a complementary fashion, thereby resulting in chopped ac voltage at the output. The fundamental or rms voltage can be regulated by varying the time ratio, $\tau = t_{on} / (t_{on} + t_{off})$ of the chopping transistors, where t_{on} and t_{off} are the time duration when the switch is closed and opened respectively. In early years, one of the primary hurdles in widespread deployment of this technology was practical realization of a bidirectional switch using Bipolar Junction Transistor (BJT) based devices. A three-phase version of the ac chopper consisted of six bidirectional switches which required twelve BJTs as shown in figure 2.3 [4].



Figure 2.3: Simplified circuit schematic of three-phase AC chopper with six bidirectional switches realized by twelve bipolar junction transistors

The principles of single-phase full ac chopper circuit as in figure 2.2 may be extended to the three phase version as shown in figure 2.3. Since star and delta connected loads are mutually convertible, and the mode of operation is the same in either case, only a star connected load is shown in figure 2.3. The circuit consists of three BJT based bidirectional switches in series and three in shunt. The series switches either act as ac chopping switches in active mode or act as closed switches in the feedback mode. The shunt switches provide a current freewheeling path in active mode or serve as open switches if corresponding line voltage is positive.

2.4. Classical PWM Three Phase AC Chopper Circuit

An equivalent three-phase ac chopper that employed six Integrated Gate Bipolar Transistors (IGBTs) was introduced in [5]. As may be seen from figure 2.4, this topology consists of three floated wye-connected shunt IGBTs which together with their anti-parallel diodes, form bidirectional switches between any two terminals.



Figure 2.4: Simplified circuit schematic of three-phase AC chopper circuit realized with six insulated gate bipolar transistors.

The three main modes of operation of the ac chopper shown in figure 2.4, are active mode, bypass mode and freewheeling mode. In the active mode of operation, the switches S_1 , S_2 , and S_3 are closed and the rest are opened. Thus, the input voltages appear across the load, while the three line currents are conducted by one or two top switches (S_1 , S_2 and S_3) and two or one diodes antiparallel to top switches. The bypass mode of operation is the transition mode from active mode to freewheeling mode of operation. In this mode, the input and output currents maintain continuity through snubber circuitry (bypass capacitors with parallel bleeding resistors connected in shunt with each line). In the

freewheeling mode of operation, the bottom switches (S_4 , S_5 and S_6) are turned on while the rest are turned off. Thus, the three load voltages and the three input line currents are zero, while the three load currents are conducted by one bottom switch and two bottom diodes or vice-versa. The fundamental concept of ac chopper (ac-ac buck converter) topology shown in figure 2.4 can be extended to other forms of switched mode power converters such as boost converter, buck-boost converter, Ćuk converter etc. and has been reported in [6] and [7].

2.5. Vector Switching Converter

The converter shown in figure 2.4 that was originally conceived for a single source, single load system has been further generalized for N sources and M loads with a topology named Vector Switching Converter (VeSC) [8]. A simplified schematic of generalized three-phase VeSC applied to an N source and M load system is shown in figure 2.5.



Figure 2.5: Simplified circuit schematic of Vector Switching Converter (VeSC).

As may be seen from figure 2.5, one requires 3xNxM IGBTs to realize a vector switching converter that can interface N stiff three-phase voltage sources with M three-phase loads. The main motivation behind this converter is to regulate the fraction of the power drawn by each three-phase load from each of the three-phase AC sources. It may be observed that the function of each three-pole switch can be seen as selectively connecting the three phase pole current vector to the various three-phase throw voltage vectors, and varying the duty ratio of each throw to modulate the power transfer among them. These converters are termed as Vector Switching Converters, since the working principle of this converter is based on controlling the connectivity between various three-phase AC voltage or current vectors by switching among their components concurrently. It may be noted that a conventional matrix converter would require 9xNxM bidirectional switches or 18xNxM IGBTs for the same purpose [8]. However matrix converter allows significantly higher flexibility in the synthesis of output waveform in terms of its frequency and phase. While as vector switching converters are restricted to power conversion at fundamental frequency and can generate only synchronous voltages, matrix converters are capable of frequency conversion as well as synthesizing voltages that are asynchronous with the source voltage.

2.6. Matrix Converter

With the introduction of matrix converter, research on direct ac-ac power conversion without reactive components received a major boost. The matrix converter [9] has been introduced to obtain sinusoidal waveforms at both input and output ports. It has been used to incorporate bi-directionality in the converter circuit, implement independent control over input and output frequency as well as achieve variable output wave amplitude and phase. A conventional matrix converter interfacing a three phase voltage source to a three phase load is shown in figure 2.6. It may be observed that it requires 9 bidirectional or 9 IGBTs to realize the matrix converter. Thus, a matrix converter that interfaces N three phase voltage sources to M three phase loads requires 9xNxM bidirectional switches.



Figure 2.6: Simplified circuit schematic of a three phase to three phase matrix converter With nine bi-directional switches the matrix converter can theoretically assume 512 (29) different switching states combinations. But not all of them can be usefully employed. Regardless of the control method used, the choice of the matrix converter switching states combinations (also referred as matrix converter configurations) to be used must comply with two basic rules. As the converter is supplied by a voltage source and usually feeds an inductive load, the input phases should not be short-circuited and the output currents should not be interrupted. From a practical point of view these rules imply that one and

only one bi-directional switch per output phase must be switched on at any instant. Taking this constraint into account, there are 27 permitted switching combinations in a three phase to three phase matrix converter. The switching sequence of the matrix converter has been designed such that each output may be fed with a cyclic permutation of the input lines. This cyclic permutation also ensures that if the input three phase system is balanced, the three output phases are naturally equidisplaced.

2.7. Phase and Amplitude Controllable Transformer

The expansion of the applicability of ac choppers beyond voltage amplitude regulation to additional control of phase and frequency of output voltage has been accomplished by the introduction of the concept of dual virtual quadrature sources [10]. Previously, ac-ac direct power converters had been designed to be controlled by traditional duty cycle (varying between 0 and 1). Thus output voltage (V_0) was governed by a simple scalar product of duty ratio (D) and input ac voltage ($V_i \cos \theta$) as shown in equation (1).

$$V_{o} = D V_{i} \cos\theta \tag{1}$$

By introducing even harmonic terms in the duty ratio, output voltage can contain an additional quadrature component along with odd harmonics as shown in equations (2) and (3).

If
$$D = D_0 + D_2 \sin 2\theta + D_4 \sin 4\theta + D_6 \sin 6\theta + \dots$$
 (2)

then,
$$Vo = D_{ref}V_i \cos(\theta + \alpha) + D_3V_i \sin 3\theta + D_5V_i \sin 5\theta + \dots$$
 (3)

The third harmonic components get trapped in a three-wire system, and thus one can now regulate the phase and frequency content of output voltage to obtain desired fundamental and harmonic reference by controlling the amount of even harmonics introduced in the



Figure 2.7: Simplified circuit schematic of a Phase Amplitude Controllable Transformer A circuit schematic of PACT has been shown in figure 2.7. The PACT utilizes the principle of creating dual virtual quadrature sources that facilitates the generation of output voltages with controllable phase and/or harmonic levels without requiring the use of stored energy or additional sources and switches.

2.8. Multilevel Direct AC Converter

The concept of dual quadrature sources based duty ratio control has been extended to build Thin AC Converters (TACC) [2]. This TACC has been used to achieve dynamic control of the grid. Main control objective of TACC is different than inverter or matrix converter, which controlled the output voltage. TACC may be used to reflect modified asset characteristics on the line side, while using the asset within its design specifications. While the possibility of realizing AC-AC direct power conversion with single devices has been documented extensively in literature, employing this technology to utility scale voltages remains challenging. This is because of the fact that devices such as IGBTs have been predominantly limited to 1700V/1200A for fast switching applications at low cost. Larger devices have now been starting to become commonly available up to 3.3kV, but with slower switching speeds and at higher costs. On the other hand, 6.6kV IGBTs have been more recently introduced and have not been widely available in commercial markets. The ability to series connect these devices reliably has been studied, however it results in further lowering of switching speeds. Although several multilevel dc/ac converter topologies are well known, extending them to direct ac-ac conversion [23]-[27] has not been widely explored. Recently, a topology shown in figure 2.8 has been presented as multilevel ac-ac converter in [11]. Using this multilevel ac-ac converter, TACC can be implemented for higher grid voltages.



Figure 2.8: Simplified circuit schematic of multilevel ac-ac converter.

As may be seen from figure 2.8, a three-level ac-ac converter is constructed by cascading two ac-ac converters. Topologically similar to the flying capacitor multilevel

dc-ac inverter [12], [24], the goal is to clamp the voltage across each bidirectional switch by an intermediate capacitor. However, one of the challenges associated with this topology is voltage balancing and firm control of this floating capacitor. Hence, a voltage balancing transformer is proposed to mitigate this issue as shown in figure 2.8.

It may be observed from figure 2.8 that the multilevel ac-ac converter topology allows the employment of lower voltage rated IGBTs for building the bidirectional switches, thereby also reducing the switching losses to improve the converter efficiency. But the disadvantage of this converter topology lies in the fact that the number of passive components has increased with the introduction of the voltage balancing transformer. This also has led to significant increase in the size of the converter. Hence one could conclude that the direct ac-ac converters at higher voltages are a challenge to implement, due to lack of higher voltage rated semiconductor devices.

CHAPTER 3 : CENTER-POINT-CLAMPED AC-AC BUCK CONVERTER

3.1. Introduction

This chapter discusses a novel topology for ac-ac direct power conversion that is capable of providing access to the mid-voltage point on the source side. This arrangement offers a means to clamp the voltage across bidirectional switches at half the magnitude of source voltage, thereby allowing employment of low voltage power electronic devices to operate with higher system voltages. Unlike flying capacitor arrangement shown in figure 2.6, the proposed structure requires no transformer for capacitor balancing and follows principle of neutral-point-clamped dc-ac inverter introduced in [13].

Although it is tempting to term the proposed power circuit as an ac-ac 'multilevel' converter, careful examination reveals that contrary to multilevel dc-ac inverter, there are no multiple ac levels which add up to synthesize the output ac voltage. Instead, this converter functions to switch either between full input voltage wave ($v_{in}\cos\theta$) and half input voltage wave ($v_{in}/2\cos\theta$), or between half input voltage wave ($v_{in}/2\cos\theta$) and zero(0) to realize the desired output ac voltage waveform. Hence this converter topology is termed as Center-Point-Clamped AC-AC Converter, and not multilevel ac-ac converter.

3.2. Evolution of Center-Point-Clamped AC-AC Converter



Figure 3.1: Simplified circuit schematics showing single phase ac-ac converter with two bidirectional switches.



Figure 3.2: Simplified circuit schematics showing single phase ac-ac converter with four bidirectional switches.

This section discusses the evolution of the Center-Point-Clamped AC-AC Converter. Simplified circuit schematic of an ac-ac converter topology with two bidirectional switches, S_1 and S_2 is shown in figure 3.1. The bidirectional switch has been realized by series connection of two back to back insulated gate bipolar transistors (IGBT) with anti-parallel diodes. The bi-directional switches, S_1 and S_2 are operated in a complementary fashion. When switch S_1 is closed, switch S_2 is open, and vice versa. A low pass L-C filter is to filter out the high frequency switching content to get back low frequency ac sinusoid at the output. It may be observed that when switch S_1 is closed, entire input voltage v_{in} is impressed across switch S_2 . Similarly, when switch S_2 is closed, voltage across switch S_1 is v_{in} . Thus, maximum voltage stress on switches S_1 and S_2 is the peak of input voltage, v_{in} . In order to employ power devices with limited
blocking voltage capability in a system that exceeds these voltage ratings, one has to resort to series connection of switches.

In this chapter, the design of an ac-ac converter for 4.16kV three-phase distribution system has been discussed. The phase-neutral voltage for a single phase system translates to 2400V ac rms (or 3400V phase-neutral peak voltage) (v_{in}). To ensure that the devices work reliably within their Safe Operating Area (SOA), it is required to employ either a 6.6kV IGBT or a series connection of two 3.3kV devices for this application. As 6.6kV devices are not widely available, our only alternative has been to employ series connection of two 3.3kV IGBTs. This is shown in figure 3.2, where each bidirectional switch is composed of four 3.3kV IGBTs.

An ac-ac converter topology equivalent to the one in figure 3-1(b) is shown in figure 3.3. As may be seen from the simplified schematic of the single phase ac-ac converter in figure 3.3, the four bidirectional switches has been rearranged.



Figure 3.3 Simplified circuit schematic of single phase ac-ac converter.

As is well-known, it is challenging to ensure that the voltages across two series connected devices are distributed evenly during steady state and dynamic conditions. Solving this problem is one of the motivations for this thesis, and the first step towards realizing the Center-Point-Clamped AC-AC Converter topology is rearrangement of four bidirectional devices as shown in figure 3.3 to form a symmetrical converter topology. As may be seen from figure 3.3, the bidirectional switches S₁ and S₄, S₂ and S₃ are series connected to withstand higher voltage ratings at the input. The switches, S₁, S₄, and S₂, S₃ are operated in a complementary manner. When switches S₁ and S₄ are closed, entire input voltage is connected to the load. Whereas when switches S₂ and S₃ are closed, the source is disconnected and load current freewheels. In figure 3.3, inductor (L_1), inductor (L_2) and capacitor (C) are filter components. Load is represented by resistor (R).

3.3. Operating Principle of Center-Point-Clamped AC-AC Buck Converter

In order to achieve even distribution of voltage stresses on the four bidirectional switches under all operating conditions, a center point is constructed on the source side with two capacitors and is tied to the mid-point between switches S_2 and S_3 . This center-point provides an access to the midpoint of the input source voltage. Complete circuit of the proposed center point clamped ac-ac converter is shown in figure 3.4. The operating principle of this center-point-clamped ac-ac converter [37] can be thought of as similar to neutral-point-clamped dc-ac inverter [13] in the sense that output can be clamped to the center point of the input.



Figure 3.4 Simplified circuit schematic of single phase center point clamped ac-ac converter.

However, instead of generating three 'dc' levels at the pole output ($-V_{dc}/2$, 0, $+V_{dc}/2$) as obtained in a conventional neutral-point-clamped inverter, realizing three distinct 'differential ac' outputs is the key point that distinguishes this topology. It may be observed that when switches S₁, S₄ are closed and switches S₂, S₃ are open, resultant output voltage is equal to input voltage v_{in} . However, because the mid-point between switches S₂ and S₃ is tied to center point on the source, the distribution of voltage between these two open switches S₂ and S₃ is identical and is equal to $v_{in}/2$. Similarly, when switches S₂, S₃ are closed and switches S₁, S₄ are open, the resultant output voltage is zero (0). Again, the distribution of voltage between two open switches S₁ and S₄ is identical and is equal to $v_{in}/2$ because of clamp to the center point. Finally, when switches S₁ and S₃ are closed, the resultant output voltage is half the input voltage i.e. $v_{in}/2$. It may be noted that another combination with switches S₂ and S₄ closed gives the same result of voltage $v_{in}/2$ across the load. However, the difference between these two operating conditions is the capacitor that gets connected to the load. Similar to the

classical neutral-point-clamped dc-ac inverter, these two states are alternately employed so as to evenly balance the voltages across the two capacitors. Switching pattern to generate an ac waveform of desired magnitude at the load side is presented in Table 1.

Command		
v_{oref}	between 0 and	between $v_{in}/2$
(desired	$v_{in}/2$	and v_{in}
output)		
State 1	$S_2 S_3$ closed	S ₁ S ₄ closed
	$S_1 S_4$ open	$S_2 S_3$ open
State 2	S ₂ S ₄ closed	S ₂ S ₄ closed
	$S_1 S_3$ open	$S_1 S_3$ open
State 3	$S_2 S_3$ closed	S ₁ S ₄ closed
	$S_1 S_4$ open	$S_2 S_3$ open
State 4	$S_1 S_3$ closed	$S_1 S_3$ closed
	$S_2 S_4$ open	$S_2 S_4$ open

Table 1. Switching pattern of the proposed center-point-clamped ac-ac converter

One of the unique features of the proposed converter topology is the method to balance input side capacitor voltage for achieving consistent center-point voltage, $v_{in}/2$. As may be seen from Table 1, when switches S₂ and S₃ are closed for zero output voltage level in State 1, capacitors C_{i1} and C_{i2} get equally charged. In order to achieve output voltage $v_{in}/2$ in State 2, switches S₂ and S₄ are closed. This results in discharging of C_{i1} and charging of C_{i2}. For the next zero output voltage level in State 3, switches S₂ and S₃ are closed as in State 1. However, for State 4, instead of S₂ and S₄ for $v_{in}/2$, we close S₁ and S₃. Thus, in this state C_{i2} gets discharged whereas C_{i1} gets charged. This alternate charging and discharging sequence of capacitors C_{i1} and C_{i2} ensures input capacitor voltage balancing. This methodology of capacitor voltage balancing is also applied when the output voltage v_{oref} is regulated from $v_{in}/2$ to v_{in} , as may be seen from Table 1.

^{3.4.} Methodology for Balancing Voltages across Input capacitors

CHAPTER 4 : CIRCUIT DESIGN OF CENTER-POINT-CLAMPED AC-AC BUCK CONVERTER

4.1. Introduction

This chapter discusses the design considerations for power circuit design of a center point clamped ac-ac converter. Power circuit design involves determination of the inductance and capacitance values in the converter system. These reactive elements may be represented as functions of inductor ripple current, capacitor ripple voltage and switching frequency. Values of reactive elements are determined using inductor volt-sec balance and capacitor current charge balance methods. Analysis and design methods applied to ac-ac converter are found to be similar to the corresponding dc-dc converter due to their design and functional similarities.

4.2. Circuit Description



Figure 4.1: Simplified circuit schematic of a single phase of center-point-clamped ac-ac buck converter.

A simplified circuit schematic of a single phase of center-point-clamped ac-ac buck converter is shown in figure 4.1. A representative center-point-clamped ac-ac buck converter is designed for 4.16kV/1MW three-phase application. For a single phase system as shown in figure 4.1, this translates to 2400V ac rms phase-neutral input voltage (or 3400V phase-neutral peak voltage) (v_{in}). The output voltage (v_o) is desired to be set within 10% to 90% of v_{in} . This converter uses 4 bidirectional switches S₁, S₂, S₃, and S₄ which are realized with 8 Insulated Gate Bipolar Transistors (IGBTs) that are switched at 1200Hz (f_{sw}). Filter components include input capacitors C_{i1} and C_{i2} , output inductor (L) and output capacitor (C). R_{Ci1} , R_{Ci2} , R_L and R_C are parasitic resistances associated with these filter components and the load is represented by a resistor R. We neglect the dynamics associated with input filter capacitors C_{i1} , C_{i2} , and their parasitic resistances R_{Ci1} and R_{Ci2} . Hence it is assumed that input voltage is split evenly between the two filter capacitors at any given time.

When switches S_1 and S_4 are closed, and switches S_2 and S_3 are open, output voltage is v_{in} . When switches S_1 and S_4 are open, and switches S_2 and S_3 are closed, output voltage is 0. And lastly, when switches S_1 and S_3 (or S_2 and S_4) are closed and switches S_2 and S_4 (or S_1 and S_3) are open, output voltage is $v_{in}/2$. Thus when the desired output voltage is less than 50% of v_{in} , then the converter is switched between 0 and $v_{in}/2$. Whereas, when the desired output voltage is more than 50% of v_{in} , then the converter is switched between $v_{in}/2$ and v_{in} .

4.3. Power Circuit Design

Let us consider 2 regions, in region1 desired output voltage ranges from 0 to $v_{in}/2$, and in region 2 desired output voltage ranges from $v_{in}/2$ to v_{in} . Region 1: (desired v_o ranges from 0 to $v_{in}/2$):

Interval T_1 (switches S_1 , S_3 (or S_2 , S_4) are closed and S_2 , S_4 (or S_1 , S_3) are open) as shown in figure 4.2.





(a) Switches S_1 , S_3 are closed and S_2 , S_4 (b) Switches S_2 , S_4 are closed and S_1 , S_3 are open. are open.

Figure 4.2: Equivalent circuit of center-point-clamped ac-ac converter in interval T₁.

Applying Kirchoff's voltage laws, we get

$$\frac{v_{in}}{2} - v_L - i_L R_L - v_o = 0 \tag{4.1}$$

Therefore,

$$v_L = \frac{v_{in}}{2} - i_L R_L - v_o \tag{4.2}$$

Current through capacitor,

$$i_c = i_L - \frac{v_o}{R} \tag{4.3}$$

In interval T_2 switches S_2 , S_3 are closed and switches S_1 , S_4 are open. The equivalent circuit of center point clamped ac-ac converter in interval T_2 is shown in figure 4.3.



Figure 4.3: Equivalent circuit of center-point-clamped ac-ac converter in interval T₂. Applying Kirchoff's voltage laws, we get

$$v_L + i_L R_L + v_o = 0 (4.4)$$

Therefore,

$$v_L = -i_L R_L - v_o \tag{4.5}$$

Current through capacitor,

$$i_c = i_L - \frac{v_o}{R} \tag{4.6}$$

Let the time period be *T* and *d* be the duty ratio. Thus, *dT* is the time when switches S_2 , S_4 are closed and also when switches S_1 , S_3 are closed. For interval T_1 , let the duration of operation be 2*dT*. In interval T_2 , duration of operation will be,

$$T - 2dT = (1 - 2d)T (4.7)$$

$$T = \frac{1}{f_{sw}} \tag{4.8}$$

 f_{sw} is the switching frequency.

In steady-state operation, average inductor voltage and average capacitor current over time period T must be zero.

Average inductor voltage,

$$\langle v_L \rangle = (\frac{v_{in}}{2} - i_L R_L - v_o) 2d + (-i_L R_L - v_o)(1 - 2d)$$
 (4.9)

$$(2d)\frac{v_{in}}{2} - v_o - i_L R_L = 0 \tag{4.10}$$

Average capacitor current,

$$\langle i_c \rangle = (i_L - \frac{v_o}{R})2d + (i_L - \frac{v_o}{R})(1 - 2d)$$
 (4.11)

$$i_L - \frac{v_o}{R} = 0$$
 (4.12)

Substituting equation (4.12) in (4.10)

$$(2d)\frac{v_{in}}{2} - v_o - \left(\frac{v_o}{R}\right)R_L = 0 \tag{4.13}$$

$$(2d)\frac{v_{in}}{2} - v_o \left(1 + \left(\frac{R_L}{R}\right)\right) = 0 \tag{4.14}$$

We get,

$$\frac{v_o}{v_{in}} = \frac{d}{\left(\frac{R_L}{R} + 1\right)} \tag{4.15}$$

Since $R_L \ll R$,

$$v_0 = d v_{in} \tag{4.16}$$

In interval T_1 , from equation (4.1),

 $R_L \ll R$, we can neglect the voltage drop across the parasitic resistance of inductor.

So,
$$v_L = (\frac{v_{in}}{2} - v_o)$$
 (4.17)

31

$$L\frac{di_L}{dt} = \left(\frac{v_{in}}{2} - v_o\right) \tag{4.18}$$

$$\Delta i_L = \frac{(\frac{v_{in}}{2} - v_o)}{L} (2d)T$$
(4.19)

peak to peak $\Delta i_L = \frac{\left(\frac{v_{in}}{2} - v_o\right)}{L} (2d)T$ (4.20)

 Δi_L is taken as 10% of inductor current i_L , we can get i_L , from equation (4.12)

Therefore,
$$L = \frac{\left(\frac{\nu_{in}}{2} - \nu_o\right)}{\Delta i_L} (2d)T$$
 (4.21)

total charge deposited on capacitor plates Q is given by,

$$Q = \Delta v_C C \tag{4.22}$$

Since R_C is negligible, v_C is equal to v_o . Δv_c is taken as 10% of capacitor voltage v_o .

Total charge Q is the area of triangle of Δi_L waveform [13],

$$Q = \frac{1}{2} \frac{\Delta i_L}{2} \frac{T}{2}$$

$$\tag{4.23}$$

Substituting equation (4.23) in (4.22)

$$C = \frac{\Delta i_L T}{8\Delta v_C} \tag{4.24}$$

Region 2: (desired v_o ranges from $v_{in}/2$ to v_{in}):

Interval T_1 (switches S_1 , S_3 (or S_2 , S_4) are closed and S_2 , S_4 (or S_1 , S_3) are open) as shown in figure 4.4.



Figure 4.4: Equivalent circuit of center-point-clamped ac-ac converter in interval T₁.

Applying Kirchoff's voltage laws, we get

$$\frac{v_{in}}{2} - v_L - i_L R_L - v_o = 0 \tag{4.25}$$

Therefore,

$$v_L = \frac{v_{in}}{2} - i_L R_L - v_o \tag{4.26}$$

Current through capacitor,

$$i_c = i_L - \frac{v_o}{R} \tag{4.27}$$

In interval T_2 , switches S_2 , S_4 are closed and switches S_1 , S_3 are open. The equivalent circuit of ac-ac converter in interval T_2 is shown in figure 4.5.



Figure 4.5: Equivalent circuit of center-point-clamped ac-ac converter in interval T₂.

Applying Kirchoff's voltage laws, we get

$$v_{in} - v_L - i_L - v_o = 0 \tag{4.28}$$

Therefore,

$$v_L = v_{in} - i_L R_L - v_o \tag{4.29}$$

Current through capacitor,

$$i_c = i_L - \frac{v_o}{R} \tag{4.30}$$

Let the time period be *T*. *dT* is the time period when switches S_2 , S_4 are closed and switches S_1 , S_3 are open. In interval T_1 , switches S_1 , S_3 (or S_2 , S_4) are closed and S_2 , S_4 (or S_1 , S_3) are open,

$$T - dT = (1 - d)T (4.31)$$

$$T = \frac{1}{f_{sw}} \tag{4.32}$$

34

In steady-state operation, average inductor voltage and average capacitor current over time period T must be zero.

Average inductor voltage

$$\langle v_L \rangle = \left(\frac{v_{in}}{2} - i_L R_L - v_o\right)(1 - d) + (v_{in} - i_L R_L - v_o) d$$
 (4.33)

$$(1+d)\frac{v_{in}}{2} - v_o - i_L R_L = 0 \tag{4.34}$$

Average capacitor current,

$$\langle i_c \rangle = (i_L - \frac{v_o}{R})D + (i_L - \frac{v_o}{R})(1 - D)$$
 (4.35)

$$i_L - \frac{v_o}{R} = 0$$
 (4.36)

Substituting equation (4.36) in (4.34)

$$(1+d)\frac{v_{in}}{2} - v_o - (\frac{v_o}{R})R_L = 0$$
(4.37)

$$(1+d)\frac{v_{in}}{2} - v_o(1+\left(\frac{R_L}{R}\right)) = 0$$
(4.38)

$$\frac{V_o}{V_{in}} = \frac{d+1}{2(\frac{R_L}{R}+1)}$$
(4.39)

In interval T_2 , from equation (4.29),

Since $R_L \ll R$, we can neglect the voltage drop across the parasitic resistance of inductor.

So,
$$v_L = v_{in} - v_o$$
 (4.40)

$$L\frac{dI_L}{dt} = v_{in} - v_o \tag{4.41}$$

35

$$\Delta i_L = \frac{v_{in} - v_o}{L} dT \tag{4.42}$$

peak to peak $\Delta i_L = \frac{v_{in} - v_o}{L} dT$ (4.43)

 Δi_L is taken as 10% of inductor current $i_{L_{i,j}}$ we can get i_L , from equation (4.36)

Therefore,

$$L = \frac{v_{in} - v_o}{\Delta i_L} dT \tag{4.44}$$

Total charge deposited on capacitor plates Q is given by,

$$Q = C\Delta v_C \tag{4.45}$$

Since R_C is negligible, v_C is equal to v_o . Δv_C is taken as 10% of capacitor voltage v_o .

Total charge Q is the area of triangle of Δi_L waveform [13]

$$Q = \frac{1}{2} \frac{\Delta i_L}{2} \frac{T}{2}$$

$$\tag{4.46}$$

Substituting equation (4.46) in (4.45)

$$C = \frac{\Delta i_L T}{8\Delta v_C} \tag{4.47}$$

Substituting values of v_o , v_{in} , d, T, Δi_L , Δv_C in equations (4.21), (4.24) (4.44), (4.47), we can get inductance and capacitance values as shown in Table 2.

Table 4.1 Inductance and capacitance values of CENTER-POINT-CLAMPED ac-ac converter.

	Desired v_o	D	L	С
Region1	$0.1 v_{\rm in}$	0.1	2.3 mH	300 µF
	$0.4 v_{in}$	0.4	9.25 mH	18.8 µF
Region 2	$0.6 v_{in}$	0.2	13.8 mH	8.34 μF
	0.9 v _{in}	0.8	20.8 mH	3.71 μF

For L > 20.8 mH, the converter enters the discontinuous conduction mode and for L < 20.8 mH, the converter cannot operate in region 2, for $v_{ref} = 0.9 v_{in}$. For $C < 300 \mu$ F, the converter cannot operate in region 1, $v_{ref} = 0.1 v_{in}$. So we select output filter components values in such a way that the converter can operate in both regions 1 and 2.

Therefore, output filter components values are,

$$L = 20.8 \text{ mH}$$
$$C = 300 \text{ }\mu\text{F}$$

CHAPTER 5 : DYNAMIC ANALYSIS OF CENTER-POINT-CLAMPED AC-AC BUCK CONVERTER

5.1. Introduction

In this chapter using a state space modelling and an average technique, transfer functions G_d (transfer function between the output voltage to duty ratio) and G_v (transfer function between the output voltage to input voltage) for the center point clamped ac-ac converter are derived. Frequency response of the system is evaluated using transfer functions G_v and G_d .

5.2. Circuit Description

A simplified circuit schematic of a single phase of center-point-clamped ac-ac converter is shown in Figure 5.1.



- Figure 5.1: Simplified circuit schematic of a single phase of center-point-clamped ac-ac buck converter.
 - A representative center-point-clamped ac-ac buck converter circuit description is

same as given in chapter 4.

5.3. State Space Equations

State space equations for a generic switched circuit that is operated in two intervals T_1 and T_2 are given by:

I. Interval T₁ (operating for duration
$$dT$$
 in a given time period T)
 $\dot{x} = A_1 x + b_1 u$ (5.1)

Interval T₂ (operating for duration (1-d)T in a given time period *T*) II.

$$\dot{x} = A_2 x + b_2 u \tag{5.3}$$

$$y = c_2^T x (5.4)$$

where *x* is the state of the system, *u* is the input, and *y* is the output [14]

State space equations in interval T_1 can be expressed with reference to *d* as:

$$\dot{x} = (A_1 x + b_1 u) d \tag{5.5}$$

State space equations in interval T_2 can be expressed with reference to (1-d) as:

$$\dot{x} = (A_2 x + b_2 u)(1 - d) \tag{5.7}$$

$$y = c_2^T x (1 - d) (5.8)$$

Averaging both sets of equations for the two switched intervals we get

$$\dot{x} = d(A_1 x + b_1 u) + (1 - d)(A_2 x + b_2 u)$$
(5.9)

$$y = dc_1^T x + (1 - d)c_2^T x (5.10)$$

._ _

5.4. Dynamic Analysis

For the center-point-clamped ac-ac buck converter circuit shown in Figure 5.1, state variables are inductor current, i_L , and capacitor voltage, v_c . Input is input voltage v_{in} , and output is output voltage v_o

Let us consider 2 regions, in region 1 desired output voltage ranges from 0 to $v_{in}/2$, and in region 2 desired output voltage ranges from $v_{in}/2$ to v_{in} .

Region 1: (desired v_o ranges from 0 to $v_{in}/2$):

Interval T_1 (switches S_1, S_3 (or S_2, S_4) are closed and S_2, S_4 (or S_1, S_3) are open) as shown in Figure 5.2.



(a) Switches S_1 , S_3 are closed and S_2 , S_4 (b) Switches S_2 , S_4 are closed and S_1 , S_3 are open. are open.

Figure 5.2: Equivalent circuits of a center-point-clamped ac-ac converter in interval T₁.

Applying Kirchoff's voltage laws, we get

$$\frac{v_{in}}{2} = i_L R_L + L \frac{di_L}{dt} + R_c (i_L - i_o) + v_c$$
(5.11)

$$v_c = (i_o - i_L)R_c + i_o R (5.12)$$

Rearranging equation 5.12,

$$i_o = \frac{v_c + i_L R_c}{R_c + R} \tag{5.13}$$

Substituting i_o in equation 5.11,

$$\frac{v_{in}}{2} = i_L R_L + L \frac{di_L}{dt} + R_c \left(i_L - \frac{v_c + i_L R_c}{R_c + R} \right) + v_c$$
(5.14)

$$\frac{v_{in}}{2} = i_L R_L + L \frac{di_L}{dt} + R_c \left(\frac{-v_c + i_L R}{R_c + R}\right) + v_c$$
(5.15)

$$\frac{\nu_{in}}{2} = i_L R_L + L \frac{di_L}{dt} + R_c \left(\frac{i_L R}{R_c + R}\right) - R_c \left(\frac{\nu_c}{R_c + R}\right) + \nu_c$$
(5.16)

$$\frac{v_{in}}{2} = i_L (R_L + R || R_c) + L \frac{di_L}{dt} + R \left(\frac{v_c}{R_c + R} \right)$$
(5.17)

Thus, state equation for i_L is,

$$\frac{di_L}{dt} = -\frac{\left(R_L + (R||R_c)\right)}{L} i_L - \frac{R}{(R+R_c)L} v_c + \frac{1}{2L} v_{in}$$
(5.18)

Applying Kirchoff's current law, we get

$$(i_L - i_o) = C \frac{dv_c}{dt}$$
(5.19)

Substituting i_o in equation 5.19,

$$\left(i_L - \frac{v_c + i_L R_c}{R_c + R}\right) = C \frac{dv_c}{dt}$$
(5.20)

Thus, state equation for v_c is,

$$\frac{dv_c}{dt} = \frac{R}{(R+R_c)C} i_L - \frac{1}{(R+R_c)C} v_C$$
(5.21)

We can combine equations 5.18 and 5.21 to give

$$\begin{bmatrix} \frac{di_L}{dt} \\ \frac{dv_c}{dt} \end{bmatrix} = \begin{bmatrix} \frac{-(R_L + (R_c \mid \mid R))}{L} & -\frac{R}{(R + R_c)L} \\ \frac{R}{(R + R_c)C} & -\frac{1}{(R + R_c)C} \end{bmatrix} \begin{bmatrix} i_L \\ v_c \end{bmatrix} + \begin{bmatrix} \frac{1}{2L} \\ 0 \end{bmatrix} v_{in}$$
(5.22)

Finally, output equation for v_o is,

$$[v_o] = \begin{bmatrix} R_c \mid \mid R & \frac{R}{R+R_c} \end{bmatrix} \begin{bmatrix} i_L \\ v_c \end{bmatrix}$$
(5.23)

Interval T_2 (switches S_1, S_4 are open and switches S_2, S_3 are closed) as shown in figure 5.3:



Figure 5.3: Equivalent circuit of a center-point-clamped ac-ac converter in interval T_2 .

Applying Kirchoff's voltage laws, we get

$$0 = i_L R_L + L \frac{di_L}{dt} + R_c (i_L - i_o) + v_c$$
(5.24)

$$v_c = (i_o - i_L)R_c + i_o R \tag{5.25}$$

Rearranging equation 5.25,

$$i_o = \frac{v_c + i_L R_c}{R_c + R} \tag{5.26}$$

Substituting i_o in equation 5.24,

$$0 = i_L R_L + L \frac{di_L}{dt} + R_c \left(i_L - \frac{v_c + i_L R_c}{R_c + R} \right) + v_c$$
(5.27)

$$0 = i_L R_L + L \frac{di_L}{dt} + R_c \left(\frac{-v_c + i_L R}{R_c + R}\right) + v_c$$
(5.28)

$$0 = i_L R_L + L \frac{di_L}{dt} + R_c \left(\frac{i_L R}{R_c + R}\right) - R_c \left(\frac{v_c}{R_c + R}\right) + v_c$$
(5.29)

$$0 = i_L(R_L + R||R_c) + L\frac{di_L}{dt} + R\left(\frac{v_c}{R_c + R}\right)$$
(5.30)

Thus, state equation for i_L is,

$$\frac{di_L}{dt} = -\frac{\left(R_L + (R||R_c)\right)}{L} i_L - \frac{R}{(R+R_c)L} v_c$$
(5.31)

Applying Kirchoff's current law, we get

$$(i_L - i_o) = C \frac{dv_c}{dt}$$
(5.32)

Substituting i_o in equation 5.32,

$$\left(i_L - \frac{v_c + i_L R_c}{R_c + R}\right) = C \frac{dv_c}{dt}$$
(5.33)

Thus, state equation for v_c is,

$$\frac{dv_c}{dt} = \frac{R}{(R+R_c)c} i_L - \frac{1}{(R+R_c)c} v_c$$
(5.34)

We can combine equations 5.31 and 5.34 to give

$$\begin{bmatrix} \frac{di_L}{dt} \\ \frac{dv_c}{dt} \end{bmatrix} = \begin{bmatrix} \frac{-(R_L + (R_c \mid \mid R))}{L} & -\frac{R}{(R + R_c)L} \\ \frac{R}{(R + R_c)C} & -\frac{1}{(R + R_c)C} \end{bmatrix} \begin{bmatrix} i_L \\ v_c \end{bmatrix} + \begin{bmatrix} 0 \\ 0 \end{bmatrix} v_{in}$$
(5.35)

Finally, output equation for v_o is,

$$[v_o] = \begin{bmatrix} R_c \mid \mid R & \frac{R}{R+R_c} \end{bmatrix} \begin{bmatrix} i_L \\ v_c \end{bmatrix}$$
(5.36)

Thus, from equations 5.22 and 5.23, we get

$$A_{1} = \begin{bmatrix} \frac{-(R_{L} + (R_{c} | |R))}{L} & -\frac{R}{(R + R_{c})L} \\ \frac{R}{(R + R_{c})C} & -\frac{1}{(R + R_{c})C} \end{bmatrix}$$
(5.37)

$$b_1 = \begin{bmatrix} \frac{1}{2L} \\ 0 \end{bmatrix} \tag{5.38}$$

$$c_1^T = \begin{bmatrix} R_c \mid \mid R & \frac{R}{R+R_c} \end{bmatrix}$$
(5.39)

And, from equations 5.35 and 5.36, we get

$$A_{2} = \begin{bmatrix} \frac{-(R_{L} + (R_{c} ||R))}{L} & -\frac{R}{(R + R_{c})L} \\ \frac{R}{(R + R_{c})C} & -\frac{1}{(R + R_{c})C} \end{bmatrix}$$
(5.40)

$$b_2 = \begin{bmatrix} 0\\0 \end{bmatrix} \tag{5.41}$$

$$c_2^T = \begin{bmatrix} R_c \mid \mid R & \frac{R}{R+R_c} \end{bmatrix}$$
(5.42)

Finally, from equations 5.9 and 5.10, we obtain the matrices

$$A = \begin{bmatrix} \frac{-(R_L + (R_c ||R))}{L} & -\frac{R}{(R + R_c)L} \\ \frac{R}{(R + R_c)C} & -\frac{1}{(R + R_c)C} \end{bmatrix}$$
(5.43)

$$b = \begin{bmatrix} \frac{d}{2L} \\ 0 \end{bmatrix}$$
(5.44)

$$c^{T} = \begin{bmatrix} (R_{c} \mid \mid R) & \frac{R}{R+R_{c}} \end{bmatrix}$$
(5.45)

Thus the dynamic equations for the center-point-clamped ac-ac buck converter system for Region 1 are:

$$\begin{bmatrix} \frac{di_L}{dt} \\ \frac{dv_c}{dt} \end{bmatrix} = \begin{bmatrix} \frac{-(R_L + (R_c \mid \mid R))}{L} & -\frac{R}{(R + R_c)L} \\ \frac{R}{(R + R_c)C} & -\frac{1}{(R + R_c)C} \end{bmatrix} \begin{bmatrix} i_L \\ v_c \end{bmatrix} + \begin{bmatrix} \frac{d}{2L} \\ 0 \end{bmatrix} v_{in}$$
(5.46)

$$[v_o] = \begin{bmatrix} R_c \mid \mid R & \frac{R}{R+R_c} \end{bmatrix} \begin{bmatrix} i_L \\ v_c \end{bmatrix}$$
(5.47)

Now let us introduce a perturbation in input voltage (u) and duty ratio (d) causing a corresponding perturbation in state variables (x) and output (y).

$$x = X + \hat{x} \tag{5.48}$$

$$y = Y + \hat{y} \tag{5.49}$$

$$u = U + \hat{u} \tag{5.50}$$

$$d = D + \hat{d} \tag{5.51}$$

Substituting in equations 5.46 and 5.47,

$$\begin{bmatrix} \frac{dI_L}{dt} + \frac{d\hat{\iota}_L}{dt} \\ \frac{dV_C}{dt} + \frac{d\hat{v}_C}{dt} \end{bmatrix} = \begin{bmatrix} \frac{-(R_L + (R_C \mid \mid R))}{L} & -\frac{R}{(R + R_C)L} \\ \frac{R}{(R + R_C)C} & -\frac{1}{(R + R_C)C} \end{bmatrix} \begin{bmatrix} I_L + \hat{\iota}_L \\ V_C + \hat{v}_C \end{bmatrix} + \begin{bmatrix} \frac{D + \hat{d}}{2L} \\ 0 \end{bmatrix} (V_{in} + \hat{v}_{in})$$
(5.52)

$$\begin{bmatrix} v_o + \widehat{v_o} \end{bmatrix} = \begin{bmatrix} R_c \mid \mid R \quad \frac{R}{R+R_c} \end{bmatrix} \begin{bmatrix} I_L + \widehat{\iota_L} \\ V_c + \widehat{v_c} \end{bmatrix}$$
(5.53)

But, from equations 5.46 and 5.47,

$$\begin{bmatrix} \frac{dI_L}{dt} \\ \frac{dV_C}{dt} \end{bmatrix} = \begin{bmatrix} \frac{-(R_L + (R_C \mid \mid R))}{L} & -\frac{R}{(R + R_C)L} \\ \frac{R}{(R + R_C)C} & -\frac{1}{(R + R_C)C} \end{bmatrix} \begin{bmatrix} I_L \\ V_C \end{bmatrix} + \begin{bmatrix} \frac{D}{2L} \\ 0 \end{bmatrix} V_{in}$$
(5.54)

$$[V_o] = \begin{bmatrix} R_c \mid \mid R & \frac{R}{R+R_c} \end{bmatrix} \begin{bmatrix} I_L \\ V_c \end{bmatrix}$$
(5.55)

Thus subtracting equation 5.54 from 5.52, and equation 5.55 from 5.53, we get

$$\begin{bmatrix} \frac{d\hat{\iota}_L}{dt} \\ \frac{d\hat{\nu}_C}{dt} \end{bmatrix} = \begin{bmatrix} \frac{-(R_L + (R_c \mid \mid R))}{L} & -\frac{R}{(R + R_c)L} \\ \frac{R}{(R + R_c)C} & -\frac{1}{(R + R_c)C} \end{bmatrix} \begin{bmatrix} \hat{\iota}_L \\ \hat{\nu}_C \end{bmatrix} + \begin{bmatrix} \frac{\hat{d}}{2L} \\ 0 \end{bmatrix} V_{in} + \begin{bmatrix} \frac{D}{2L} \\ 0 \end{bmatrix} \hat{\nu}_{in}$$
(5.56)

$$\left[\widehat{\upsilon_{o}}\right] = \left[R_{c} \mid\mid R \quad \frac{R}{R+R_{c}}\right] \left[\widehat{\upsilon_{c}}\right]$$
(5.57)

Transforming equation (5.56) to Laplace domain we get,

$$sL\hat{\iota}_{L}(s) = -(R_{L} + R \mid \mid R_{C})\hat{\iota}_{L}(s) - \frac{R}{(R+R_{C})}\hat{v}_{C}(s) + \frac{D}{2}\hat{v}_{in}(s) + \frac{V_{in}}{2}\hat{d}(s)$$
(5.58)

$$sC\widehat{v_C}(s) = \frac{R}{R+R_C}\widehat{\iota_L}(s) - \frac{1}{R+R_C}\widehat{v_C}(s)$$
(5.59)

Rearranging,

$$(s\mathcal{C} + \frac{1}{R+R_C})\widehat{v_C}(s) = \frac{R}{R+R_C}\widehat{\iota_L}(s)$$
(5.60)

$$\left(sC\left(\frac{R+R_{C}}{R}\right) + \frac{1}{R}\right)\widehat{v_{C}}(s) = \widehat{\iota_{L}}(s)$$
(5.61)

Similarly, transforming output equation (5.57) to Laplace domain we get

$$\widehat{v_o}(s) = (R \mid\mid R_C) \widehat{\iota_L}(s) + \frac{R}{(R+R_C)} \widehat{v_C}(s)$$
(5.62)

$$\widehat{v_o}(s)\left(\frac{R+R_C}{R}\right) - R_C \widehat{\iota_L}(s) = \widehat{v_C}(s)$$
(5.63)

Substituting from equation (5.61),

$$\widehat{v_o}(s)\left(\frac{R+R_C}{R}\right) - R_C(sC(\frac{R+R_C}{R}) + \frac{1}{R})\,\widehat{v_C}(s) = \widehat{v_C}(s) \tag{5.64}$$

$$\widehat{v_o}(s)\left(\frac{R+R_C}{R}\right) = R_C\left(sC\left(\frac{R+R_C}{R}\right) + \frac{1}{R}\right)\,\widehat{v_C}(s) + \widehat{v_C}(s) \tag{5.65}$$

$$\widehat{v_o}(s) = R_C \left(sC + \frac{1}{R + R_C} \right) \widehat{v_C}(s) + \left(\frac{R}{R + R_C} \right) \widehat{v_C}(s)$$
(5.66)

$$\widehat{v_o}(s) = sR_c C \ \widehat{v_c}(s) + \widehat{v_c}(s) \tag{5.67}$$

$$\widehat{v_o}(s) = (1 + sR_cC)\,\widehat{v_c}(s) \tag{5.68}$$

Substituting in equation (5.61),

$$\left(sC\left(\frac{R+R_{C}}{R}\right) + \frac{1}{R}\right)\widehat{v}_{C}(s) = \widehat{\iota}_{L}(s)$$
(5.69)

$$\frac{1}{1+sR_CC}\left(sC\left(\frac{R+R_C}{R}\right) + \frac{1}{R}\right)\widehat{\nu_o}(s) = \widehat{\iota_L}(s)$$
(5.70)

Substituting in equation (5.58),

$$sL\frac{1}{1+sR_{c}C}\left(sC\left(\frac{R+R_{c}}{R}\right)+\frac{1}{R}\right)\widehat{v_{o}}(s) = -(R_{L}+R\mid\mid R_{C})\frac{1}{1+sR_{c}C}\left(sC\left(\frac{R+R_{c}}{R}\right)+\frac{1}{R}\right)\widehat{v_{o}}(s) - \frac{R}{(R+R_{c})}\frac{1}{1+sR_{c}C}\widehat{v_{o}}(s) + \frac{D}{2}\widehat{v_{in}}(s) + \frac{V_{in}}{2}\widehat{d}(s)$$
(5.71)

$$sL\frac{1+sC(R+R_{C})}{1+sR_{C}C}\frac{1}{R}\widehat{v_{o}}(s) = -(R_{L}+R||R_{C})\frac{1+sC(R+R_{C})}{1+sR_{C}C}\frac{1}{R}\widehat{v_{o}}(s) - \frac{R}{(R+R_{C})}\frac{1}{1+sR_{C}C}\widehat{v_{o}}(s) + \frac{D}{2}\widehat{v_{in}}(s) + \frac{V_{in}}{2}\widehat{d}(s)$$
(5.72)

$$sL\frac{1+sC(R+R_{C})}{1+sR_{C}C}\frac{1}{R}\widehat{v_{o}}(s) + (R_{L}+R||R_{C})\frac{1+sC(R+R_{C})}{1+sR_{C}C}\frac{1}{R}\widehat{v_{o}}(s) + \frac{R}{(R+R_{C})}\frac{1}{1+sR_{C}C}\widehat{v_{o}}(s) = \frac{D}{2}\widehat{v_{in}}(s) + \frac{V_{in}}{2}\widehat{d}(s)$$
(5.73)

$$sL\frac{1+sC(R+R_{C})}{R}\,\widehat{v_{o}}(s) + (R_{L}+R\mid|R_{C})\frac{1+sC(R+R_{C})}{R}\,\widehat{v_{o}}(s) + \frac{R}{(R+R_{C})}\,\widehat{v_{o}}(s) = (1+sR_{C}C)(\frac{D}{2}\,\widehat{v_{in}}(s) + \frac{V_{in}}{2}\,\hat{d}(s))$$
(5.74)

$$(sL + R_L + R ||R_C) \frac{1 + sC(R + R_C)}{R} \,\widehat{v_o}(s) + \frac{R}{(R + R_C)} \,\widehat{v_o}(s) = (1 + sR_CC)(\frac{D}{2}\,\widehat{v_{in}}(s) + \frac{V_{in}}{2}\,\hat{d}(s))$$
(5.75)

$$\left[\left(sL + R_L + R \mid \mid R_C\right) \frac{1 + sC(R + R_C)}{R} + \frac{R}{(R + R_C)}\right] \widehat{v_0}(s) = (1 + sR_CC) \left(\frac{D}{2}\widehat{v_{in}}(s) + \frac{V_{in}}{2}\widehat{d}(s)\right)$$
(5.76)

From equation (5.76), we get transfer functions as,

$$G_d = \frac{\widehat{v_o(s)}}{\widehat{d}(s)}_{\widehat{v_{in}}(s)=0}$$
(5.77)

$$G_{\nu} = \frac{\widehat{v_0(s)}}{\widehat{v_{in}(s)}}_{\hat{d}(s)=0}$$
(5.78)

where G_d is the transfer function between the output voltage to duty ratio and G_v is the transfer function between the output voltage to input voltage.

Substituting equation (5.76) in (5.77) and (5.78), we obtain,

$$G_{d} = \frac{V_{in}}{2(\left[\left(\frac{Cs\left(\frac{R+R_{C}}{R}\right)+\frac{1}{R}}{1+sCR_{C}}\right)(Ls+R_{L}+R \mid \mid R_{C})\right]+\frac{R}{(R+R_{C})})}$$
(5.79)

$$G_{v} = \frac{D}{2(\left[\left(\frac{Cs\left(\frac{R+R_{C}}{R}\right)+\frac{1}{R}}{1+sCR_{C}}\right)(Ls+R_{L}+R\mid|R_{C})\right]+\frac{R}{(R+R_{C})})}$$
(5.80)

For desired output voltage (0 to $v_{in}/2$):

Transfer functions G_d and G_v are given by equations (5.79) and (5.80). Uncompensated bode plots of equation (5.79) and equation (5.80) are shown in Figure (5.4) and Figure (5.5),



Figure 5.4: Magnitude and phase plots of transfer function G_d for output voltage range 0 to $v_{in}/2$.



Figure 5.5: Magnitude and phase plots of transfer function G_v for output voltage range 0

to $v_{in}/2$.

Region 2: (desired v_o ranges from $v_{in}/2$ to v_{in}):

Interval T_1 (switches S_1, S_4 are closed and switches S_2, S_3 are open) as shown in figure 5.6



Figure 5.6: Equivalent circuit of a center-point-clamped ac-ac buck converter in interval T_1 .

Applying Kirchoff's voltage laws, we get

$$v_{in} = i_L R_L + L \frac{di_L}{dt} + R_c (i_L - i_o) + v_c$$
(5.81)

$$v_c = (i_o - i_L)R_c + i_o R \tag{5.82}$$

Rearranging equation 5.82,

$$i_o = \frac{v_c + i_L R_c}{R_c + R} \tag{5.83}$$

Substituting i_o in equation 5.81,

$$v_{in} = i_L R_L + L \frac{di_L}{dt} + R_c \left(i_L - \frac{v_c + i_L R_c}{R_c + R} \right) + v_c$$
(5.84)

$$v_{in} = i_L R_L + L \frac{di_L}{dt} + R_c \left(\frac{-v_c + i_L R}{R_c + R}\right) + v_c$$
(5.85)

$$v_{in} = i_L R_L + L \frac{di_L}{dt} + R_c \left(\frac{i_L R}{R_c + R}\right) - R_c \left(\frac{v_c}{R_c + R}\right) + v_c$$
(5.86)

$$v_{in} = i_L(R_L + R||R_c) + L\frac{di_L}{dt} + R\left(\frac{v_c}{R_c + R}\right)$$
(5.87)

Thus, state equation for i_L is,

$$\frac{di_L}{dt} = -\frac{\left(R_L + (R||R_c)\right)}{L} i_L - \frac{R}{(R+R_c)L} v_c + \frac{1}{L} v_{in}$$
(5.88)

Applying Kirchoff's current law, we get

$$(i_L - i_o) = C \frac{dv_c}{dt}$$
(5.89)

Substituting i_o in equation 5.89,

$$\left(i_L - \frac{v_c + i_L R_c}{R_c + R}\right) = C \frac{dv_c}{dt}$$
(5.92)

Thus, state equation for v_c is,

$$\frac{dv_c}{dt} = \frac{R}{(R+R_c)C} i_L - \frac{1}{(R+R_c)C} v_C$$
(5.93)

We can combine equations 5.88 and 5.93 to give

$$\begin{bmatrix} \frac{di_L}{dt} \\ \frac{dv_c}{dt} \end{bmatrix} = \begin{bmatrix} \frac{-(R_L + (R_c \mid \mid R))}{L} & -\frac{R}{(R + R_c)L} \\ \frac{R}{(R + R_c)C} & -\frac{1}{(R + R_c)C} \end{bmatrix} \begin{bmatrix} i_L \\ v_c \end{bmatrix} + \begin{bmatrix} \frac{1}{L} \\ 0 \end{bmatrix} v_{in}$$
(5.94)

Finally, output equation for v_o is,

$$[v_o] = \begin{bmatrix} R_c \mid \mid R & \frac{R}{R+R_c} \end{bmatrix} \begin{bmatrix} i_L \\ v_c \end{bmatrix}$$
(5.95)

Interval T_2 (switches S_1, S_3 (or S_2, S_4) are closed and S_2, S_4 (or S_1, S_3) are open) as shown in figure 5.7.



Figure 5.7: Equivalent circuits of a center-point-clamped ac-ac buck converter in interval T_2 .

Applying Kirchoff's voltage laws, we get

$$\frac{v_{in}}{2} = i_L R_L + L \frac{di_L}{dt} + R_c (i_L - i_o) + v_c$$
(5.96)

$$v_c = (i_o - i_L)R_c + i_o R \tag{5.97}$$

Rearranging equation 5.97,

$$i_o = \frac{v_c + i_L R_c}{R_c + R} \tag{5.98}$$

Substituting i_o in equation 5.96,

$$\frac{v_{in}}{2} = i_L R_L + L \frac{di_L}{dt} + R_c \left(i_L - \frac{v_c + i_L R_c}{R_c + R} \right) + v_c$$
(5.99)

$$\frac{v_{in}}{2} = i_L R_L + L \frac{di_L}{dt} + R_c \left(\frac{-v_c + i_L R}{R_c + R}\right) + v_c$$
(5.100)

$$\frac{\nu_{in}}{2} = i_L R_L + L \frac{di_L}{dt} + R_c \left(\frac{i_L R}{R_c + R}\right) - R_c \left(\frac{\nu_c}{R_c + R}\right) + \nu_c$$
(5.101)

$$\frac{v_{in}}{2} = i_L(R_L + R||R_c) + L\frac{di_L}{dt} + R\left(\frac{v_c}{R_c + R}\right)$$
(5.102)

Thus, state equation for i_L is,

$$\frac{di_L}{dt} = -\frac{\left(R_L + (R||R_c)\right)}{L} i_L - \frac{R}{(R+R_c)L} v_c + \frac{1}{2L} v_{in}$$
(5.103)

Applying Kirchoff's current law, we get

$$(i_L - i_o) = C \frac{dv_c}{dt}$$
(5.104)

Substituting i_o in equation 5.104,

$$\left(i_L - \frac{v_c + i_L R_c}{R_c + R}\right) = C \frac{dv_c}{dt}$$
(5.105)

Thus, state equation for v_c is,

$$\frac{dv_c}{dt} = \frac{R}{(R+R_c)c} i_L - \frac{1}{(R+R_c)c} v_c$$
(5.106)

We can combine equations 5.103 and 5.106 to give

$$\begin{bmatrix} \frac{di_L}{dt} \\ \frac{dv_c}{dt} \end{bmatrix} = \begin{bmatrix} \frac{-(R_L + (R_c \mid \mid R))}{L} & -\frac{R}{(R + R_c)L} \\ \frac{R}{(R + R_c)C} & -\frac{1}{(R + R_c)C} \end{bmatrix} \begin{bmatrix} i_L \\ v_c \end{bmatrix} + \begin{bmatrix} \frac{1}{2L} \\ 0 \end{bmatrix} v_{in}$$
(5.107)

Finally, output equation for v_o is,

$$[v_o] = \begin{bmatrix} R_c \mid \mid R & \frac{R}{R+R_c} \end{bmatrix} \begin{bmatrix} i_L \\ v_c \end{bmatrix}$$
(5.108)

Thus, from equations 5.94 and 5.95, we get

$$A_{1} = \begin{bmatrix} \frac{-(R_{L} + (R_{c} ||R))}{L} & -\frac{R}{(R + R_{c})L} \\ \frac{R}{(R + R_{c})C} & -\frac{1}{(R + R_{c})C} \end{bmatrix}$$
(5.109)

$$b_1 = \begin{bmatrix} \frac{1}{L} \\ 0 \end{bmatrix} \tag{5.110}$$

$$c_1^T = \begin{bmatrix} R_c \mid \mid R & \frac{R}{R+R_c} \end{bmatrix}$$
(5.111)

And, from equations 5.107 and 5.108, we get

$$A_{2} = \begin{bmatrix} \frac{-(R_{L} + (R_{c} ||R))}{L} & -\frac{R}{(R + R_{c})L} \\ \frac{R}{(R + R_{c})C} & -\frac{1}{(R + R_{c})C} \end{bmatrix}$$
(5.112)

$$b_2 = \begin{bmatrix} \frac{1}{2L} \\ 0 \end{bmatrix}$$
(5.113)

$$c_2^T = \begin{bmatrix} R_c \mid \mid R & \frac{R}{R+R_c} \end{bmatrix}$$
(5.114)

Finally, from equations 5.9 and 5.10, we obtain the matrices

$$A = \begin{bmatrix} \frac{-(R_L + (R_C ||R))}{L} & -\frac{R}{(R + R_C)L} \\ \frac{R}{(R + R_C)C} & -\frac{1}{(R + R_C)C} \end{bmatrix}$$
(5.115)

$$b = \begin{bmatrix} \frac{d+1}{2L} \\ 0 \end{bmatrix}$$
(5.116)

$$c^{T} = \begin{bmatrix} (R_{c} \mid \mid R) & \frac{R}{R+R_{c}} \end{bmatrix}$$
(5.117)

Thus the dynamic equations for the center-point-clamped ac-ac converter system for region 2 are:

$$\begin{bmatrix} \frac{di_L}{dt} \\ \frac{dv_c}{dt} \end{bmatrix} = \begin{bmatrix} \frac{-(R_L + (R_c ||R))}{L} & -\frac{R}{(R + R_c)L} \\ \frac{R}{(R + R_c)C} & -\frac{1}{(R + R_c)C} \end{bmatrix} \begin{bmatrix} i_L \\ v_c \end{bmatrix} + \begin{bmatrix} \frac{d+1}{2L} \\ 0 \end{bmatrix} v_{in}$$
(5.118)

$$[v_o] = \begin{bmatrix} R_c \mid \mid R & \frac{R}{R+R_c} \end{bmatrix} \begin{bmatrix} i_L \\ v_c \end{bmatrix}$$
(5.119)

Now let us introduce a perturbation in input voltage (u) and duty ratio (d) causing a corresponding perturbation in state variables (x) and output (y).

$$x = X + \hat{x} \tag{5.120}$$

$$y = Y + \hat{y} \tag{5.121}$$

$$u = U + \hat{u} \tag{5.122}$$

$$d = D + \hat{d} \tag{5.123}$$

Substituting in equations 5.118 and 5.119,

$$\begin{bmatrix} \frac{dI_L}{dt} + \frac{d\hat{I}_L}{dt} \\ \frac{dV_C}{dt} + \frac{d\hat{V_C}}{dt} \end{bmatrix} = \begin{bmatrix} \frac{-(R_L + (R_C \mid \mid R))}{L} & -\frac{R}{(R + R_C)L} \\ \frac{R}{(R + R_C)C} & -\frac{1}{(R + R_C)C} \end{bmatrix} \begin{bmatrix} I_L + \hat{I}_L \\ V_C + \hat{V_C} \end{bmatrix} + \begin{bmatrix} \frac{D + \hat{d} + 1}{2L} \\ 0 \end{bmatrix} (V_{in} + \hat{V_{in}})$$
(5.124)

55

$$\begin{bmatrix} V_o + \widehat{v_o} \end{bmatrix} = \begin{bmatrix} R_c \mid \mid R \quad \frac{R}{R+R_c} \end{bmatrix} \begin{bmatrix} I_L + \widehat{\iota}_L \\ V_c + \widehat{v_c} \end{bmatrix}$$
(5.125)

But, from equations 5.118 and 5.119,

$$\begin{bmatrix} \frac{dI_L}{dt} \\ \frac{dV_C}{dt} \end{bmatrix} = \begin{bmatrix} \frac{-(R_L + (R_C \mid \mid R))}{L} & -\frac{R}{(R + R_C)L} \\ \frac{R}{(R + R_C)C} & -\frac{1}{(R + R_C)C} \end{bmatrix} \begin{bmatrix} I_L \\ V_C \end{bmatrix} + \begin{bmatrix} \frac{D+1}{2L} \\ 0 \end{bmatrix} V_{in}$$
(5.126)

$$[V_o] = \begin{bmatrix} R_c \mid \mid R & \frac{R}{R+R_c} \end{bmatrix} \begin{bmatrix} I_L \\ V_c \end{bmatrix}$$
(5.127)

Thus subtracting equation (5.126) from (5.124), and equation (5.127) from (5.125), we

get

$$\begin{bmatrix} \frac{dt_{L}}{dt} \\ \frac{dv_{C}}{dt} \end{bmatrix} = \begin{bmatrix} \frac{-(R_{L} + (R_{c} \mid \mid R))}{L} & -\frac{R}{(R+R_{c})L} \\ \frac{R}{(R+R_{c})C} & -\frac{1}{(R+R_{c})C} \end{bmatrix} \begin{bmatrix} \hat{t}_{L} \\ \hat{v}_{C} \end{bmatrix} + \begin{bmatrix} \frac{\hat{d}}{2L} \\ 0 \end{bmatrix} V_{in} + \begin{bmatrix} \frac{D+1}{2L} \\ 0 \end{bmatrix} \hat{v}_{in}$$
(5.128)

$$\left[\widehat{\nu_{o}}\right] = \left[R_{c} \mid \mid R \quad \frac{R}{R+R_{c}}\right] \left[\widehat{\nu_{c}}\right]$$
(5.129)

Transforming equation (5.128) to Laplace domain we get,

$$sL\hat{\iota}_{L}(s) = -(R_{L} + R \mid \mid R_{C})\hat{\iota}_{L}(s) - \frac{R}{(R+R_{C})}\hat{\nu}_{C}(s) + \frac{D+1}{2}\hat{\nu}_{in}(s) + \frac{V_{in}}{2}\hat{d}(s)$$
(5.130)

$$sC\widehat{\nu_C}(s) = \frac{R}{R+R_C}\widehat{\iota_L}(s) - \frac{1}{R+R_C}\widehat{\nu_C}(s)$$
(5.131)

Rearranging,

$$(sC + \frac{1}{R+R_C})\widehat{\nu_C}(s) = \frac{R}{R+R_C}\widehat{\iota_L}(s)$$
(5.132)

56

$$\left(sC\left(\frac{R+R_{C}}{R}\right) + \frac{1}{R}\right)\widehat{\nu_{C}}(s) = \widehat{\iota_{L}}(s)$$
(5.133)

Similarly, transforming output equation (5.129) to Laplace domain we get

$$\widehat{v_o}(s) = (R \mid\mid R_C)\widehat{\iota_L}(s) + \frac{R}{(R+R_C)}\widehat{v_C}(s)$$
(5.134)

$$\widehat{v_o}(s)\left(\frac{R+R_C}{R}\right) - R_C \widehat{\iota_L}(s) = \widehat{v_C}(s)$$
(5.135)

Substituting from equation (5.133),

$$\widehat{v_o}(s)\left(\frac{R+R_C}{R}\right) - R_C(sC(\frac{R+R_C}{R}) + \frac{1}{R})\,\widehat{v_C}(s) = \widehat{v_C}(s) \tag{5.136}$$

$$\widehat{\nu_o}(s)\left(\frac{R+R_C}{R}\right) = R_C\left(sC\left(\frac{R+R_C}{R}\right) + \frac{1}{R}\right)\,\widehat{\nu_C}(s) + \widehat{\nu_C}(s) \tag{5.137}$$

$$\widehat{\nu_o}(s) = R_C \left(sC + \frac{1}{R + R_C} \right) \widehat{\nu_C}(s) + \left(\frac{R}{R + R_C} \right) \widehat{\nu_C}(s)$$
(5.138)

$$\widehat{v_o}(s) = sR_C C \ \widehat{v_C}(s) + \widehat{v_C}(s) \tag{5.139}$$

$$\widehat{v_o}(s) = (1 + sR_cC)\,\widehat{v_c}(s) \tag{5.140}$$

Substituting in equation (5.133),

$$\left(sC\left(\frac{R+R_{C}}{R}\right) + \frac{1}{R}\right)\widehat{\nu}_{C}(s) = \widehat{\iota}_{L}(s)$$
(5.134)

$$\frac{1}{1+sR_CC}\left(sC\left(\frac{R+R_C}{R}\right) + \frac{1}{R}\right)\widehat{\nu_o}(s) = \widehat{\iota_L}(s)$$
(5.135)

Substituting in equation (5.130),

$$57$$

$$sL\frac{1}{1+sR_{C}C}\left(sC\left(\frac{R+R_{C}}{R}\right)+\frac{1}{R}\right)\widehat{v_{o}}(s) = -(R_{L}+R \mid\mid R_{C})\frac{1}{1+sR_{C}C}\left(sC\left(\frac{R+R_{C}}{R}\right)+\frac{1}{R}\right)\widehat{v_{o}}(s) - \frac{R}{(R+R_{C})}\frac{1}{1+sR_{C}C}\widehat{v_{o}}(s) + \frac{D+1}{2}\widehat{v_{in}}(s) + \frac{V_{in}}{2}\widehat{d}(s)$$

$$sL\frac{1+sC(R+R_{C})}{1+sR_{C}C}\frac{1}{R}\widehat{v_{o}}(s) = -(R_{L}+R \mid\mid R_{C})\frac{1+sC(R+R_{C})}{1+sR_{C}C}\frac{1}{R}\widehat{v_{o}}(s) - \frac{R}{(R+R_{C})}\frac{1}{1+sR_{C}C}\widehat{v_{o}}(s) + \frac{1}{2}\sum_{k=1}^{N}\widehat{v_{k}}(s) + \frac{1}{2}\sum_{k=1}$$

$$\frac{D+1}{2}\widehat{v_{in}}(s) + \frac{V_{in}}{2}\hat{d}(s)$$
(5.137)

$$sL\frac{1+sC(R+R_{C})}{1+sR_{C}C}\frac{1}{R}\widehat{v_{0}}(s) + (R_{L}+R||R_{C})\frac{1+sC(R+R_{C})}{1+sR_{C}C}\frac{1}{R}\widehat{v_{0}}(s) + \frac{R}{(R+R_{C})}\frac{1}{1+sR_{C}C}\widehat{v_{0}}(s) = \frac{D+1}{2}\widehat{v_{in}}(s) + \frac{V_{in}}{2}\widehat{d}(s)$$
(5.138)

$$sL\frac{1+sC(R+R_{C})}{R}\widehat{v_{o}}(s) + (R_{L}+R||R_{C})\frac{1+sC(R+R_{C})}{R}\widehat{v_{o}}(s) + \frac{R}{(R+R_{C})}\widehat{v_{o}}(s) = (1+sR_{C}C)(\frac{1+D}{2}\widehat{v_{in}}(s) + \frac{V_{in}}{2}\hat{d}(s))$$
(5.139)

$$(sL + R_L + R ||R_C) \frac{1 + sC(R + R_C)}{R} \,\widehat{v_0}(s) + \frac{R}{(R + R_C)} \,\widehat{v_0}(s) = (1 + sR_CC)(\frac{D+1}{2}\,\widehat{v_{in}}(s) + \frac{V_{in}}{2}\,\hat{d}(s))$$
(5.140)

$$[(sL + R_L + R || R_C) \frac{1 + sC(R + R_C)}{R} + \frac{R}{(R + R_C)}] \widehat{v_o}(s) = (1 + sR_CC)(\frac{D+1}{2}\widehat{v_{in}}(s) + \frac{V_{in}}{2}\widehat{d}(s))$$
(5.141)

From equation (5.141), we get transfer functions as,

$$G_d = \frac{\widehat{v_o(s)}}{\widehat{d}(s)} \frac{1}{\widehat{v_{in}}(s) = 0}$$
(5.142)

$$G_{\nu} = \frac{\hat{v_0}(s)}{\hat{v_{in}}(s)}_{\hat{d}(s)=0}$$
(5.143)
where G_d is the transfer function between the output voltage to duty ratio and G_v is the transfer function between the output voltage to input voltage.

Substituting equation (5.141) in (5.142) and (5.143), we obtain,

$$G_{d} = \frac{V_{in}}{2(\left[\left(\frac{Cs\left(\frac{R+R_{C}}{R}\right)+\frac{1}{R}}{1+sCR_{C}}\right)(sL+R_{L}+R \mid \mid R_{C})\right]+\frac{R}{(R+R_{C})})}$$
(5.144)

$$G_{v} = \frac{D+1}{2(\left[\left(\frac{Cs\left(\frac{R+R_{C}}{R}\right)+\frac{1}{R}}{1+sCR_{C}}\right)(sL+R_{L}+R \mid \mid R_{C})\right]+\frac{R}{(R+R_{C})})}$$
(5.145)

For desired output voltage ($v_{in}/2$ to v_{in}):

Transfer functions G_d and G_v are given by equations (5.144) and (5.145). Uncompensated bode plots of equation (5.144) and equation (5.145) are shown in figure 5.8 and figure 5.9.



Figure 5.8: Magnitude and phase plots of transfer function G_d for output voltage range $v_{in}/2$ to v_{in} .



Figure 5.9: Magnitude and phase plots of transfer function G_v for output voltage range $v_{in}/2$ to v_{in} .

CHAPTER 6 : CONTROLLER DESIGN FOR CLOSED LOOP CONTROL OF CENTER-POINT-CLAMPED AC-AC CONVERTER

6.1. Introduction

The design of a feedback compensator for closed loop control of the center-pointclamped ac-ac buck converter has been discussed in this chapter. Transfer function G_d (transfer function between the output voltage to duty ratio) is used for designing the feedback compensator a MATLAB-SISO (Single Input/Single Output) tool. SISO tool is a GUI (Graphical User Interface) for compensator designs.

6.2. Circuit Description

A simplified circuit schematic of a single phase of center-point-clamped ac-ac



buck converter is shown in figure 6.1.

Figure 6.1: Simplified circuit schematic of a single phase of center-point-clamped ac-ac buck converter.

A representative center-point-clamped ac-ac converter circuit description is same as given in chapter 4.

6.3. Compensator Design

Using the transfer function G_d (transfer function between the output voltage to duty ratio), a feedback compensator has been designed to make the center point clamped ac-ac converter stable. MATLAB SISO tool has been used to design the feedback compensator for the converter. As may be seen from the Bode plots of the transfer function of the uncompensated converter system, the gain margin is infinity, but the phase margin is very little. In order to make the converter stable for all regions of operation, the phase margin has to be improved to at least fifty degrees. It may also be seen from the unit step input response of the uncompensated converter system that the peak amplitude rises to about 200 percent of unity, thus making the converter system underdamped. SISO tool gets updated G_d , from MATLAB workspace and produces closed loop and open loop bode plots. To improve the phase and gain margin of the bode plot, pole should either be placed/moved to the right and the zero towards left of the bode plot.

The compensator G_c is given by,

$$G_c = \frac{(1+0.023s)}{s \ (1+0.0001 \ s)} \tag{6.1}$$



Figure 6.2: Step response plot of uncompensated converter output voltage to duty cycle transfer function, G_d .

For the uncompensated transfer function G_d , the designed feedback compensator (G_c) has damped the peak overshoot, good command following performance as the system exhibits good line regulation and fast transient response. This compensator G_c is employed in the converter system as shown in figure 6.3.



Figure 6.3: Control block diagram of center-point-clamped ac-ac converter.

It may be seen from figure 6.4 that the addition of the feedback controller has improved the phase margin to 50 degrees. Also, it may be observed from figure 6.5 that the unit step input response of the transfer function, $G_d * G_c$ of the compensated converter system that the system peak overshoot has been damped.



Figure 6.4: Magnitude and phase plots of transfer function $G_d * G_c$ for output voltage range 0 to v_{in} .



Figure 6.5: Step response plot of transfer function $G_d * G_c$ for output voltage range 0 to v_{in} .

CHAPTER 7 : INPUT FILTER CONSIDERATIONS FOR CENTER-POINT-CLAMPED AC-AC CONVERTER

7.1. Introduction

The design of input filter for power converters in critical applications must adhere to the EMI (Electro-Magnetic Interference) norms for various application domains, such as defense, aerospace and telecommunication industry [28], [29]. It may be seen that an input filter may be designed that closely conforms to the harmonic and EMI standards, but it might cause significant performance degradation of the power converter itself. The performance degradation is attributed to the introduction of additional states in the system, which leads to change in the converter transfer functions as well as change in converter source impedance [30]. Thus, it may be seen that although input filter is necessary to reject harmonics in the input current due to switching in the converter as well as to protect converter from input voltage transients, yet care must be taken in designing the input filter and the corresponding controller such that it does not alter the converter performance or destabilize the converter [31]-[36]. This chapter deals with the input filter design considerations for a Center-Point-Clamped AC-AC converter. It may be seen in the following sections that the designed input filter improves the spectral performance of the converter. A feedback controller is designed which significantly improves the phase margin of the converter system.

7.2. Modified Center-Point-Clamped AC-AC Converter Circuit

Modified circuit schematic of a single phase Center-Point-Clamped AC-AC converter with the introduction of an input side inductor, L_i is shown in figure 7.1.



Figure 7.1: Simplified circuit schematic of Center-Point-Clamped AC-AC converter with input filter.

It may be seen from [31] that a switched mode converter due to its switching action feeds in discontinuous current from its input side. It may be observed that this discontinuous current contains harmonics which may be expressed as a function of the switching function. These harmonics are in the order of multiples of switching frequency. Thus, they may be filtered out using a low pass filter. As may be seen from figure 7.1, an inductor has been introduced into the modified center-point-clamped ac-ac buck converter circuit schematic to form a low pass filter with the input side capacitors, C_{i1} and C_{i2} . This filter removes the unwanted harmonic currents fed into the unregulated input power supply. But the introduction of the input filter has destabilizing effect on the converter as the source impedance gets changed. The effect in the performance and stability of the converter due to input filter is studied in the following sections.

7.3. Description of Modified Center-Point-Clamped AC-AC Buck Converter Circuit As may be seen from figure 7.1, V_s is the input voltage source, L_i is the input side inductor and R_{Li} is the parasitic resistance involved with L_i , C_{i1} and C_{i2} are the input side voltage balancing capacitors, R_{Ci1} and R_{Ci2} are the parasitic resistances involved with C_{i1} and C_{i2} , R_{i1} and R_{i2} are the input side resistances behaving as a potential divider, S_1 , S_2 , S_3 and S_4 are the four bidirectional switches realized by IGBTs, L_o is the output side inductor and R_{Lo} is the parasitic resistance associated with it, C_o is the output side capacitor and R_{Co} is the parasitic resistance of C_o , and V_o is the voltage across the load resistor, R_o .

7.4. Dynamic Analysis of Modified Center-Point-Clamped AC-AC Buck Converter

It may be observed that the introduction of an inductor to form the input side low pass filter would change the converter transfer functions obtained in Section 5.4. It is so because the addition of the input filter introduces new states into the converter state space equations obtained in Section 5.3. Also, it may be seen from figure.7.1 that the modified converter circuit schematic incorporates the parasitic resistances involved with the components, L_i , C_{i1} , C_{i2} , L_o , and C_o . A new dynamic analysis of the modified converter circuit, as shown has been performed using state space averaging technique. Here, the dynamics involved with the input side inductor and capacitors have been taken into account. Thus, the assumption that input voltage is split evenly between the two filter capacitors at any given time has been ignored in this analysis, rather the dynamics associated with the capacitor voltages have also been examined in the analysis. It has been assumed that the converter is operating in continuous conduction mode. As already discussed in Section 5.4, the converter has two regions, in region 1 desired output voltage ranges from 0 to $v_s/2$, and in region 2 desired output voltage ranges from $v_s/2$ to v_s . Here, the state variables are input filter inductor current, i_s , input filter capacitor voltages, v_{Ci1} and v_{Ci2} , output filter inductor current, i_{Lo} , and output filter capacitor voltage, v_{Co} . The dynamic analysis is performed for both regions as shown in Section 5.4 and the state space equations derived for both the regions of converter operation are shown as follows:

Region1: (Reference voltage ranges between 0 to $v_s/2$)

$$\begin{bmatrix} \frac{di_{s}}{dt} \\ \frac{dv_{Ci1}}{dt} \\ \frac{dv_{Ci2}}{dt} \\ \frac{di_{L0}}{dt} \\ \frac{dv_{Co}}{dt} \end{bmatrix} = A_{1} * \begin{bmatrix} i_{s} \\ v_{Ci1} \\ v_{Ci2} \\ i_{L0} \\ v_{Co} \end{bmatrix} + \begin{bmatrix} 1 \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix} v_{s}$$

$$(7.1)$$

$$v_{o} = \begin{bmatrix} 0 & 0 & 0 & \frac{R_{Co} * R_{o}}{R_{Co} + R_{o}} & \frac{R_{o}}{R_{Co} + R_{o}} \end{bmatrix} \begin{bmatrix} v_{Ci1}^{*} \\ v_{Ci2} \\ i_{Lo} \\ v_{Co} \end{bmatrix}$$
(7.2)

where, A₁ is

$$\begin{bmatrix} -R_{Li} - \frac{(R_{Ci1} * R_{i1})}{(R_{Ci1} + R_{i1})} - \frac{(R_{Ci2} * R_{i2})}{(R_{Ci2} + R_{i2})} & -\frac{(R_{i1})}{(R_{Ci1} + R_{i1})} & -\frac{(R_{i2})}{(R_{Ci2} + R_{i2})} & \begin{pmatrix} \left(\frac{R_{Ci1} * R_{i1}}{(R_{Ci1} + R_{i1})} + \frac{(R_{Ci2} * R_{i2})}{2}\right) \\ 0 & \frac{(R_{i1})}{(R_{Ci1} + R_{i1})} & -\frac{(R_{i1})}{(R_{Ci2} + R_{i2})} & -\frac{(R_{i2})}{(R_{Ci2} + R_{i2})} & \begin{pmatrix} \left(\frac{R_{Ci1} * R_{i1}}{(R_{Ci1} + R_{i1})} + \frac{(R_{Ci2} * R_{i2})}{2}\right) \\ 0 & \frac{(R_{i2})}{(R_{Ci1} + R_{i1})} & -\frac{(R_{i1})}{(R_{Ci2} + R_{i2})} & -\frac{(R_{i2})}{(R_{Ci2} + R_{i2})} & \frac{(R_{i1} * R_{i1})}{(R_{Ci2} + R_{i2})} & \frac{(R_{i1} * R_{i1})}{(R_{Ci2} + R_{i2})} & \frac{(R_{i1} * R_{i1})}{(R_{Ci2} + R_{i2})} \\ \begin{pmatrix} \left(\frac{R_{Ci1} * R_{i1}}{(R_{Ci1} + R_{i1})} + \frac{(R_{Ci2} * R_{i2})}{(R_{Ci2} + R_{i2})}\right) \\ \frac{(R_{i1} * R_{i1})}{2} & \frac{(R_{i1} * R_{i1})}{(2 * (R_{Ci1} + R_{i1}))} & \frac{(R_{i2} + R_{i2})}{(2 * (R_{Ci2} + R_{i2}))} \\ \end{pmatrix} \\ \end{pmatrix} & = \frac{R_{L0} - \left(R_{L0} - \left(\frac{R_{L0} * R_{10}}{(R_{Ci1} + R_{i1})} + \frac{(R_{Ci2} * R_{i2})}{(R_{Ci2} + R_{i2})}\right) - \frac{(R_{co} * R_{o})}{(R_{co} + R_{o})} \\ -\frac{(R_{co} * R_{o})}{(R_{co} + R_{o})} \\ \end{pmatrix} \right] \\ \end{pmatrix}$$

$$\begin{bmatrix} \frac{di_s}{dt} \\ \frac{dv_{Ci1}}{dt} \\ \frac{dv_{Ci2}}{dt} \\ \frac{di_{LO}}{dt} \\ \frac{dv_{CO}}{dt} \end{bmatrix} = A_2 * \begin{bmatrix} i_s \\ v_{Ci1} \\ v_{Ci2} \\ i_{LO} \\ v_{CO} \end{bmatrix} + \begin{bmatrix} \frac{1}{L_i} \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix} v_s$$

$$(7.3)$$

$$v_{o} = \begin{bmatrix} 0 & 0 & 0 & \frac{R_{Co} * R_{o}}{R_{Co} + R_{o}} & \frac{R_{o}}{R_{Co} + R_{o}} \end{bmatrix} \begin{bmatrix} v_{c} \\ v_{Ci1} \\ v_{Ci2} \\ i_{Lo} \\ v_{Co} \end{bmatrix}$$
(7.4)

$$\begin{pmatrix} -R_{Ll} - \frac{(R_{Cl1} * R_{l1})}{(R_{Cl1} + R_{l1})} - \frac{(R_{Cl2} * R_{l2})}{(R_{Cl2} + R_{l2})} & -\frac{(R_{l1})}{(R_{Cl1} + R_{l1})} & -\frac{(R_{l2})}{(R_{Cl2} + R_{l2})} & \begin{pmatrix} (1 - d) * \frac{(\frac{(R_{Cl1} * R_{l1})}{2} + \frac{(R_{Cl2} * R_{l2})}{2})}{(R_{Cl1} + R_{l1})} & 0 & 0 \\ \frac{(R_{l1})}{(R_{Cl1} + R_{l1})} & -\frac{(R_{l1})}{(R_{Cl1} + R_{l1})} & -\frac{(R_{l2})}{(R_{Cl2} + R_{l2})} & \begin{pmatrix} (1 - d) * \frac{(\frac{(R_{Cl1} * R_{l1})}{2} + \frac{(R_{Cl2} + R_{l2})}{2})}{0 & 0 & 0 \\ \frac{(R_{l2})}{(R_{Cl2} + R_{l2})} & 0 & \frac{((1 - d) * R_{l1})}{(R_{Cl1} + R_{l1})} & 0 \\ 0 & \frac{((1 - d) * R_{l2})}{(R_{Cl1} + R_{l1})} & \frac{((1 - d) * R_{l2})}{(R_{Cl2} + R_{l2})} & \frac{((1 - d) * R_{l2})}{(2 * (R_{Cl1} + R_{l1}))} & \frac{((1 - d) * R_{l2})}{(2 * (R_{Cl1} + R_{l1}))} & \frac{((1 - d) * R_{l2})}{(2 * (R_{Cl1} + R_{l1}))} & \frac{(R_{Cl} * R_{l2})}{(R_{Cl} + R_{l2})} \\ \begin{pmatrix} (1 - d) * \frac{(\frac{(R_{Cl1} * R_{l1})}{(R_{Cl1} + R_{l1})} + \frac{(R_{Cl2} * R_{l2})}{(R_{Cl2} + R_{l2})}) \\ 0 & \frac{(1 - d) * (R_{Cl} * R_{l1})}{(R_{Cl1} + R_{l1})} + \frac{(R_{Cl2} * R_{l2})}{(R_{Cl} + R_{l2})} & -\frac{(R_{Co} * R_{o})}{(R_{Co} + R_{o})} \\ \end{pmatrix} \\ \end{pmatrix} \\ \begin{pmatrix} 0 & 0 & 0 & \frac{(R_{cl1} * R_{l1})}{(R_{cl1} + R_{l1})} + \frac{(R_{cl2} * R_{l2})}{(R_{cl2} + R_{l2})} \\ 0 & -R_{L0} - \left((1 - d) * \frac{(R_{cl1} * R_{l1})}{(R_{cl1} + R_{l1})} + \frac{(R_{cl2} * R_{l2})}{(R_{co} + R_{o})} \right) - \frac{(R_{co} * R_{o})}{(R_{co} + R_{o})} \\ -\frac{(1)}{(R_{co} + R_{o})} & -\frac{(1)}{(R_{co} + R_{o})} \\ \end{pmatrix} \\ \end{pmatrix}$$

Introducing a perturbation in input voltage, v_s , input side inductor current, i_s , input side capacitor voltages, v_{Ci1} and v_{Ci2} , duty ratio d_i output side inductor current, i_{Lo} and output side capacitor voltage, v_{Co} causes a corresponding perturbation in state variables i_s , v_{Ci1} , v_{Ci2} , i_{Lo} , and v_{Co} , and output v_o , as follows:

$$v_s = V_s + \hat{v}_s \tag{7.5}$$

$$i_s = I_s + \hat{\iota}_s \tag{7.6}$$

 $v_{Ci1} = V_{Ci1} + \hat{v_{Ci1}}$ (7.7)

$$v_{Ci2} = V_{Ci2} + \widehat{v_{Ci2}} \tag{7.8}$$

$$d = D + \hat{d} \tag{7.9}$$

<u>Region2:</u> (Reference voltage ranges between $v_s/2$ to v_s):

$$i_{Lo} = I_{Lo} + \widehat{\iota_{Lo}} \tag{7.10}$$

69

$$v_{Co} = V_{Co} + \widehat{v_{Co}} \tag{7.11}$$

$$v_o = V_o + \widehat{v_o} \tag{7.12}$$

where the capitalized variables represent the steady state values and the variables in hat (^) representation are the small perturbations introduced in steady state parameters. Approximating that perturbations are negligible as compared to the steady state values, small signal state space equations for center-point-clamped ac-ac converter circuit are

Region1: (Reference voltage ranges between 0 to $v_{\rm s}/2$)

$$\begin{bmatrix} \frac{d(I_{s}+\hat{i}_{s})}{dt} \\ \frac{d(V_{Ci1}+\tilde{v}_{Ci1})}{dt} \\ \frac{d(V_{Ci2}+\tilde{v}_{Ci2})}{dt} \\ \frac{d(I_{Lo}+\hat{i}_{Lo})}{dt} \\ \frac{d(V_{Lo}+\tilde{v}_{Co})}{dt} \\ \frac{d(V_{Co}+\tilde{v}_{Co})}{dt} \end{bmatrix} = \left(A_{1}+\widehat{A_{1}}\right) * \begin{bmatrix} I_{s}+\hat{i}_{s} \\ V_{Ci1}+\tilde{v}_{Ci2} \\ I_{Lo}+\tilde{i}_{Lo} \\ V_{Co}+\tilde{v}_{Co} \end{bmatrix} + \begin{bmatrix} \frac{1}{L_{i}} \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix} \left(V_{s}+\hat{v}_{s}\right)$$
(7.13)
$$(V_{o}+\hat{v}_{o}) = \begin{bmatrix} 0 & 0 & 0 & \frac{R_{Co}*R_{o}}{R_{Co}+R_{o}} & \frac{R_{o}}{R_{Co}+R_{o}} \end{bmatrix} \begin{bmatrix} I_{s}+\hat{i}_{s} \\ V_{Ci1}+\tilde{v}_{Ci1} \\ V_{Ci2}+\tilde{v}_{Ci2} \\ I_{Lo}+\tilde{v}_{Ci2} \end{bmatrix}$$
(7.14)

$$(V_{o} + \widehat{v_{o}}) = \begin{bmatrix} 0 & 0 & 0 & \frac{R_{Co} * R_{o}}{R_{Co} + R_{o}} & \frac{R_{o}}{R_{Co} + R_{o}} \end{bmatrix} \begin{bmatrix} I_{s} + \widehat{t}_{s} \\ V_{Ci1} + \widehat{v_{Ci1}} \\ V_{Ci2} + \widehat{v_{Ci2}} \\ I_{Lo} + \widehat{t_{Lo}} \\ V_{Co} + \widehat{v_{Co}} \end{bmatrix}$$
(7.14)

where
$$\widehat{A_1}$$
 is

<u>Region2:</u> (Reference voltage ranges between $v_s/2$ to v_s)

$$\begin{bmatrix} \frac{d(l_{s}+\hat{l}_{s})}{dt} \\ \frac{d(V_{Ci1}+\hat{v_{Ci1}})}{dt} \\ \frac{d(V_{Ci2}+\hat{v_{Ci2}})}{dt} \\ \frac{d(l_{Lo}+\hat{l_{Lo}})}{dt} \\ \frac{d(l_{Lo}+\hat{l_{Lo}})}{dt} \\ \frac{d(V_{Co}+\hat{v_{Co}})}{dt} \end{bmatrix} = \left(A_{2} + \widehat{A_{2}}\right) * \begin{bmatrix} I_{s} + \hat{l}_{s} \\ V_{Ci1} + \hat{v_{Ci1}} \\ V_{Ci2} + \hat{v_{Ci2}} \\ I_{Lo} + \hat{l_{Lo}} \\ V_{Co} + \hat{v_{Co}} \end{bmatrix} + \begin{bmatrix} \frac{1}{l_{i}} \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix} \left(V_{s} + \hat{v}_{s}\right)$$
(7.15)

$$(V_{o} + \widehat{v_{o}}) = \begin{bmatrix} 0 & 0 & 0 & \frac{R_{Co} * R_{o}}{R_{Co} + R_{o}} & \frac{R_{o}}{R_{Co} + R_{o}} \end{bmatrix} \begin{bmatrix} I_{s} + I_{s} \\ V_{Ci1} + \widehat{v_{Ci1}} \\ V_{Ci2} + \widehat{v_{Ci2}} \\ I_{Lo} + \widehat{\iota_{Lo}} \\ V_{Co} + \widehat{v_{Co}} \end{bmatrix}$$
(7.16)

where $\widehat{A_2}$ is

Upon elimination of the steady state terms from equations, (7.13), (7.14), (7.15) and (7.16) and neglecting the second order differential terms as they are assumed to be very negligible, the new set of state space equations are

<u>Region1:</u> (Reference voltage ranges between 0 to $v_s/2$)

$$\begin{bmatrix} \frac{d(\hat{l}_{S})}{dt} \\ \frac{d(\hat{v}_{C11})}{dt} \\ \frac{d(\hat{v}_{C12})}{dt} \\ \frac{d(\hat{v}_{C0})}{dt} \\ \frac{d(\hat{v}_{C0})}{dt} \\ \frac{d(\hat{v}_{C0})}{dt} \end{bmatrix} = (A_{1}) * \begin{bmatrix} \hat{l}_{S} \\ \hat{v}_{C11} \\ \hat{v}_{C12} \\ \hat{l}_{L0} \\ \hat{v}_{C0} \end{bmatrix} + (\widehat{A_{1}}) * \begin{bmatrix} I_{S} \\ V_{C11} \\ V_{C12} \\ I_{L0} \\ V_{C0} \end{bmatrix} + \begin{bmatrix} \frac{1}{L_{i}} \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix} (\hat{v}_{S})$$
(7.17)

$$(\widehat{\nu_{o}}) = \begin{bmatrix} 0 & 0 & 0 & \frac{R_{Co} * R_{o}}{R_{Co} + R_{o}} & \frac{R_{o}}{R_{Co} + R_{o}} \end{bmatrix} \begin{bmatrix} \widehat{l_{s}} \\ \widehat{\nu_{C11}} \\ \widehat{\nu_{C12}} \\ \widehat{l_{Lo}} \\ \widehat{\nu_{Co}} \end{bmatrix}$$
(7.18)

where
$$\widehat{A_1}$$
 is

<u>Region2:</u> (Reference voltage ranges between $v_s/2$ to v_s)

$$\begin{bmatrix} \frac{d(\hat{l}_{S})}{dt} \\ \frac{d(\hat{v}_{\widehat{C}(1)})}{dt} \\ \frac{d(\hat{v}_{\widehat{C}(12)})}{dt} \\ \frac{d(\hat{v}_{\widehat{L}(2)})}{dt} \\ \frac{d(\hat{v}_{\widehat{L}(2)})}{dt} \\ \frac{d(\hat{v}_{\widehat{C}(2)})}{dt} \end{bmatrix} = (A_{2}) * \begin{bmatrix} \hat{l}_{S} \\ \hat{v}_{\widehat{C}(1)} \\ \hat{v}_{\widehat{L}(2)} \\ \hat{v}_{\widehat{C}(2)} \end{bmatrix} + (\widehat{A}_{2}) * \begin{bmatrix} I_{S} \\ V_{C11} \\ V_{C12} \\ I_{L0} \\ V_{C0} \end{bmatrix} + \begin{bmatrix} \frac{1}{L_{i}} \\ 0 \\ 0 \\ 0 \end{bmatrix} (\widehat{v}_{S})$$
(7.19)

$$(\widehat{v_o}) = \begin{bmatrix} 0 & 0 & 0 & \frac{R_{Co} * R_o}{R_{Co} + R_o} & \frac{R_o}{R_{Co} + R_o} \end{bmatrix} \begin{bmatrix} \widehat{v_{Ci1}} \\ \widehat{v_{Ci2}} \\ \widehat{l_{Lo}} \\ \widehat{v_{Co}} \end{bmatrix}$$
(7.20)

where $\widehat{A_2}$ is

Now, transforming the above time domain equations, (7-17), (7-18), (7-19) and (7-20) into Laplace domain, we have,

<u>Region1:</u> (Reference voltage ranges between 0 to $v_s/2$)

$$\begin{split} sI * \begin{bmatrix} \hat{l}_{s} \\ \hat{v}_{C11} \\ \hat{v}_{C12} \\ \frac{l}{l_{O}} \\ \hat{v}_{C0} \end{bmatrix} &= (A_{1}) * \begin{bmatrix} \hat{l}_{s} \\ \hat{v}_{C11} \\ \hat{v}_{C12} \\ \frac{l}{l_{O}} \\ \hat{v}_{C0} \end{bmatrix} + (\widehat{A}_{1}) * \begin{bmatrix} I_{s} \\ V_{C11} \\ V_{C12} \\ I_{L0} \\ V_{C0} \end{bmatrix} + \begin{bmatrix} \frac{1}{l_{i}} \\ 0 \\ 0 \\ 0 \end{bmatrix} (\widehat{v}_{s}) \\ \\ or, (sI - A_{1}) * \begin{bmatrix} \hat{l}_{s} \\ \hat{v}_{C11} \\ \hat{v}_{C12} \\ \frac{l}{l_{O}} \\ \hat{v}_{C0} \end{bmatrix} = (\widehat{A}_{1}) * \begin{bmatrix} I_{s} \\ V_{C11} \\ V_{C12} \\ I_{L0} \\ V_{C0} \end{bmatrix} + \begin{bmatrix} \frac{1}{l_{i}} \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix} (\widehat{v}_{s}) \\ \\ or, (sI - A_{1})^{-1} * (sI - A_{1}) * \begin{bmatrix} \hat{v}_{s} \\ \hat{v}_{C11} \\ \hat{v}_{C12} \\ \hat{v}_{C0} \end{bmatrix} = (sI - A_{1})^{-1} * (\widehat{A}_{1}) * \begin{bmatrix} I_{s} \\ v_{C11} \\ \hat{v}_{C12} \\ \hat{v}_{C0} \end{bmatrix} \\ + (sI - A_{1})^{-1} * \begin{bmatrix} \frac{1}{l_{i}} \\ 0 \\ 0 \\ 0 \end{bmatrix} (\widehat{v}_{s}) \\ \\ or, \begin{bmatrix} \hat{l}_{s} \\ \hat{v}_{C11} \\ \hat{v}_{C12} \\ \hat{v}_{C0} \end{bmatrix} = (sI - A_{1})^{-1} * (\widehat{A}_{1}) * \begin{bmatrix} I_{s} \\ v_{C11} \\ v_{C12} \\ \hat{v}_{C0} \end{bmatrix} \\ + (sI - A_{1})^{-1} * \begin{bmatrix} \frac{1}{l_{i}} \\ 0 \\ 0 \\ 0 \end{bmatrix} (\widehat{v}_{s}) \\ \\ (\widehat{v}_{s}) \end{bmatrix}$$
(7.21)

or,
$$\widehat{A_1} = {A_1}' * \widehat{d}$$

where,

$$A_{1}^{\prime} = \begin{bmatrix} 0 & 0 & 0 & 0 & \frac{\left(\frac{\left(R_{c11}^{*R_{i1}}\right) + \left(R_{c12}^{*R_{i2}}\right)}{\left(R_{c11}^{*R_{i1}}\right) + \left(R_{c12}^{*R_{i2}}\right)}\right) & 0 \\ 0 & 0 & 0 & \frac{R_{i1}}{2} & 0 \\ 0 & 0 & 0 & \frac{R_{i1}}{2} & 0 \\ \frac{\left(\frac{\left(\frac{\left(R_{c11}^{*R_{i1}}\right) + \left(R_{c12}^{*R_{i2}}\right)}{2}\right)}{2}\right) - \frac{\left(R_{i1}\right)}{\left(2^{*}\left(R_{c11}^{*R_{i1}}\right) - \frac{R_{i2}}{2}\right)} & \frac{R_{i2}}{\left(2^{*}\left(R_{c12}^{*R_{i2}}\right)\right)} & \frac{R_{i2}}{\left(2^{*}\left(R_{c12}^{*R_{i2}}\right) - \frac{R_{i2}}{2}\right)} & 0 \\ 0 & 0 & 0 & - \left(\frac{\left(\frac{\left(R_{c11}^{*R_{i1}}\right) + \left(R_{c12}^{*R_{i2}}\right)}{2}\right) - \frac{R_{i2}}{2}\right)}{2} & 0 \\ 0 & 0 & 0 & - \left(\frac{R_{i2}^{*}\left(R_{i1}^{*R_{i1}}\right) + \left(R_{c12}^{*R_{i2}}\right)}{2}\right) - \frac{R_{i2}^{*}\left(R_{c12}^{*R_{i2}}\right)}{2}\right) \\ 0 & 0 & 0 & 0 & - \left(\frac{R_{i1}^{*}\left(R_{i1}^{*}\right) + \left(R_{i2}^{*}R_{i2}\right)}{2}\right) & 0 \\ 0 & 0 & 0 & 0 & - \left(\frac{R_{i1}^{*}R_{i1}}{R_{c11}^{*}R_{i1}^{*}\left(R_{c12}^{*}R_{i2}\right)}\right) - \frac{R_{i2}^{*}R_{i2}^{*}R_{i2}}{2}\right) \\ 0 & 0 & 0 & 0 & 0 & - \left(\frac{R_{i1}^{*}R_{i1}^{*}R_{i1}^{*}R_{i1}}{R_{i1}^{*}R_{i1}^{*}R_{i2}^{*}R_{i2}^{*}}\right) - \frac{R_{i2}^{*}R_{i2}^{*}R_{i2}}{2}\right) \\ 0 & 0 & 0 & 0 & - \left(\frac{R_{i1}^{*}R_{i1}^{*}R_{i1}^{*}R_{i1}^{*}R_{i1}^{*}R_{i2}^{*}R_{i2}}{2}\right) \\ 0 & 0 & 0 & 0 & - \left(\frac{R_{i2}^{*}R_{i2}^{*}R_{i2}^{*}R_{i2}}{2}\right) - \frac{R_{i2}^{*}R_{i2}^{*}R_{i2}^{*}}{2}\right) - \frac{R_{i2}^{*}R_{i2}^{*}R_{i2}^{*}R_{i2}^{*}R_{i2}^{*}R_{i2}^{*}R_{i2}^{*}R_{i2}^{*}}R_{i2}^{*}R_{i2}^{*}R_{i2}^{*}R_{i2}^{*}}R_{i2}^{*}R_{i2}^$$

Using the equation (7-22) in equation (7-21), we have,

$$\begin{bmatrix} \hat{l}_{s} \\ \hat{v}_{Ci1} \\ \hat{v}_{Ci2} \\ \hat{l}_{L0} \\ \hat{v}_{Co} \end{bmatrix} = (sI - A_{1})^{-1} * (A_{1}^{/}) * \begin{bmatrix} I_{s} \\ V_{Ci1} \\ V_{Ci2} \\ I_{Lo} \\ V_{Co} \end{bmatrix} * (\hat{d}) + (sI - A_{1})^{-1} * \begin{bmatrix} \frac{1}{L_{i}} \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix} (\hat{v}_{s})$$
(7.23)

73

(7.22)

From equation (7-18), we have,

$$(\widehat{v_{o}}) = \begin{bmatrix} 0 & 0 & 0 & \frac{R_{Co} * R_{o}}{R_{Co} + R_{o}} & \frac{R_{o}}{R_{Co} + R_{o}} \end{bmatrix} \begin{bmatrix} \widehat{l}_{S} \\ \widehat{v_{Cl1}} \\ \widehat{v_{Cl2}} \\ \widehat{l_{Lo}} \\ \widehat{v_{Co}} \end{bmatrix}$$
(7.24)

Now, equating (6-23) and (6-24), we have,

$$(\widehat{v_{o}}) = \begin{bmatrix} 0 & 0 & 0 & \frac{R_{Co} * R_{o}}{R_{Co} + R_{o}} & \frac{R_{o}}{R_{Co} + R_{o}} \end{bmatrix} * (SI - A_{1})^{-1} * \begin{pmatrix} A_{1} \\ \end{pmatrix} * \begin{bmatrix} I_{s} \\ V_{Ci1} \\ V_{Ci2} \\ I_{Lo} \\ V_{Co} \end{bmatrix} * (\hat{d}) + \begin{bmatrix} 0 & 0 & 0 & \frac{R_{Co} * R_{o}}{R_{Co} + R_{o}} & \frac{R_{o}}{R_{Co} + R_{o}} \end{bmatrix} * (SI - A_{1})^{-1} * \begin{bmatrix} \frac{1}{L_{i}} \\ 0 \\ 0 \\ 0 \end{bmatrix} (\widehat{v_{s}})$$
(7.25)

From equation (7-25), the converter output voltage (\hat{v}_o) to duty ratio transfer function (\hat{d}) , G_{d1} may be found as follows

or,
$$G_{d1} = \begin{bmatrix} 0 & 0 & 0 & \frac{R_{Co} * R_o}{R_{Co} + R_o} & \frac{R_o}{R_{Co} + R_o} \end{bmatrix} * (sI - A_1)^{-1} * (A_1^{/}) * \begin{bmatrix} I_s \\ V_{Ci1} \\ V_{Ci2} \\ I_{Lo} \\ V_{Co} \end{bmatrix}$$
(7-27)

<u>Region2:</u> (Reference voltage ranges between $v_s/2$ to v_s)

On performing Laplace transformation of equation (6-19), we have

$$sI * \begin{bmatrix} \widehat{l}_{s} \\ \widehat{v}_{\widehat{c}\iota1} \\ \widehat{v}_{\widehat{c}\iota2} \\ \widehat{l}_{Lo} \\ \widehat{v}_{\widehat{c}o} \end{bmatrix} = (A_{2}) * \begin{bmatrix} \widehat{l}_{s} \\ \widehat{v}_{\widehat{c}\iota1} \\ \widehat{v}_{\widehat{c}\iota2} \\ \widehat{l}_{Lo} \\ \widehat{v}_{\widehat{c}o} \end{bmatrix} + (\widehat{A}_{2}) * \begin{bmatrix} I_{s} \\ V_{c11} \\ V_{c12} \\ I_{Lo} \\ V_{c0} \end{bmatrix} + \begin{bmatrix} \frac{1}{L_{i}} \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix} (\widehat{v}_{s})$$

$$\begin{aligned} \text{or,} & (sI - A_2) * \begin{bmatrix} \hat{l}_s \\ \overline{V_{C11}} \\ \overline{v_{C22}} \\ \frac{1}{V_{C0}} \end{bmatrix} = (\overline{A_2}) * \begin{bmatrix} I_s \\ V_{C11} \\ V_{C2} \\ I_{Lo} \\ V_{Co} \end{bmatrix} + \begin{bmatrix} 1 \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix} (\widehat{v}_s) \\ \\ \text{or,} & (sI - A_2)^{-1} * (sI - A_2) * \begin{bmatrix} \hat{l}_s \\ \overline{V_{C11}} \\ \overline{v_{C22}} \\ \frac{1}{V_{C0}} \end{bmatrix} = (sI - A_2)^{-1} * (\overline{A_2}) * \begin{bmatrix} I_s \\ V_{C11} \\ V_{C22} \\ \frac{1}{V_{C0}} \end{bmatrix} \\ & + (sI - A_2)^{-1} * \begin{bmatrix} 1 \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix} (\widehat{v}_s) \\ \\ \text{or,} & \begin{bmatrix} \hat{l}_s \\ V_{C11} \\ V_{C12} \\ \frac{1}{V_{C12}} \\ \frac{1}{V_{C12}} \end{bmatrix} = (sI - A_2)^{-1} * (\overline{A_2}) * \begin{bmatrix} I_s \\ V_{C11} \\ V_{C12} \\ \frac{1}{V_{C12}} \\ \frac{1}{V_{C12}} \\ \frac{1}{V_{C12}} \end{bmatrix} = (sI - A_2)^{-1} * (\overline{A_2}) * \begin{bmatrix} I_s \\ V_{C11} \\ V_{C12} \\ 1_{Lo} \\ V_{C0} \end{bmatrix} + (sI - A_2)^{-1} * \begin{bmatrix} 1 \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix} (\widehat{v}_s) \\ (\widehat{v}_s) \\ \\ \text{or,} & \begin{bmatrix} 1 \\ \hat{v}_{C11} \\ \frac{1}{V_{C12}} \\ \frac{1}{V_{C12}} \\ \frac{1}{V_{C12}} \end{bmatrix} = (sI - A_2)^{-1} * (\overline{A_2}) * \begin{bmatrix} I_s \\ V_{C11} \\ V_{C12} \\ I_{Lo} \\ V_{C0} \end{bmatrix} + (sI - A_2)^{-1} * \begin{bmatrix} 1 \\ 0 \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix} (\widehat{v}_s) \\ (\widehat{v}_s) \\ (\widehat{v}_s) \\ (\widehat{v}_{C1} + \widehat{v}_{C1}) \\ (\widehat{v}_{C1} + \widehat{v}_{C1} + \widehat{v}_{C1}) \\ (\widehat{v}_$$

or,
$$\widehat{A_2} = A_2^{/} * \hat{d}$$
 (7.27)

where,

Using the equation (7-27) in equation (6-26), we have,

$$\begin{bmatrix} \hat{l}_{s} \\ \hat{v}_{Cl1} \\ \hat{v}_{Cl2} \\ \hat{l}_{Lo} \\ \hat{v}_{Co} \end{bmatrix} = (sI - A_{2})^{-1} * (A_{2}^{/}) * \begin{bmatrix} I_{s} \\ V_{Cl1} \\ V_{Cl2} \\ I_{Lo} \\ V_{Co} \end{bmatrix} * (\hat{d}) + (sI - A_{2})^{-1} * \begin{bmatrix} \frac{1}{L_{i}} \\ 0 \\ 0 \\ 0 \\ 0 \end{bmatrix} (\hat{v}_{s})$$
(7.28)

From equation (7-20), we have,

$$(\widehat{v_o}) = \begin{bmatrix} 0 & 0 & 0 & \frac{R_{Co} * R_o}{R_{Co} + R_o} & \frac{R_o}{R_{Co} + R_o} \end{bmatrix} \begin{bmatrix} \widehat{\iota_s} \\ \widehat{v_{C\iota 1}} \\ \widehat{v_{C\iota 2}} \\ \widehat{\iota_{Lo}} \\ \widehat{v_{Co}} \end{bmatrix}$$
(7.29)

From equation (7.25), the converter output voltage (\hat{v}_o) to duty ratio transfer function (\hat{d}) , G_{d2} may be found as follows

$$G_{d2} = \frac{\widehat{v_0}(s)}{\widehat{d}(s)}_{\widehat{v_{in}}(s)=0}$$
(7.26)

or,
$$G_{d2} = \begin{bmatrix} 0 & 0 & 0 & \frac{R_{Co} * R_o}{R_{Co} + R_o} & \frac{R_o}{R_{Co} + R_o} \end{bmatrix} * (sI - A_2)^{-1} * (A_2^{/}) * \begin{bmatrix} I_s \\ V_{Ci1} \\ V_{Ci2} \\ I_{Lo} \\ V_{Co} \end{bmatrix}$$
 (7-27)

7.5. Input Filter Considerations for Center-Point-Clamped AC-AC Converter Circuit The effect in the performance and stability of the converter due to input filter is studied in this section. It may be seen from the converter system dynamic analysis performed in Chapter 5, the dynamics pertaining to L_i , C_{i1} , and C_{i2} has been neglected for simplicity of analysis.



Figure 7.2: Simplified circuit schematic of Center-Point-Clamped AC-AC converter with input filter.

It has been assumed that the input voltage is split evenly between the two input side filter capacitors, i.e center-point at any given time [37]. But the input filter plays an important role in filtering out the unwanted harmonic currents fed into the unregulated input power supply [31]-[36] as well as to clamp the midpoint voltage level. Also, it may be seen from [30], [31], [39], that a switched mode converter due to its switching action feeds in

discontinuous current from its input side. It may be observed that this discontinuous current contains harmonics which may be expressed as a function of switching frequency. These harmonics are in the order of multiples of switching frequency. Thus, they may be filtered out using a low pass filter. As may be seen from figure 7.1, an inductor is present in input side of the center-point-clamped ac-ac buck converter circuit schematic to form a low pass filter with input side capacitors, C_{i1} and C_{i2} . This filter removes unwanted harmonic currents fed into the unregulated input power supply. But the introduction of input filter may have a destabilizing effect on the converter as source impedance gets changed. In order to analyze the effect of input side filter in performance and stability of the converter, state variables pertaining to input side filter components, voltage across L_i, current through capacitors, C_{i1} and C_{i2} , have to be taken into account while deriving dynamic equations of converter using state space averaging technique. It may be observed that the state space equation then would involve a state space matrix of 5x5 dimensions. Hence, it may be noted that it would be really difficult to derive converter transfer functions after the introduction of perturbation in five state variables involved in the dynamic equations.

This section discusses a technique to simplify the algebra involved in deriving the converter transfer functions, which incorporates the dynamics associated with all the possible state variables viz. five state variables for the system shown in figure 7.1. It may be seen in [38], that the dynamics of input filter may be incorporated in the transfer function G_{vd} (converter duty ratio to output voltage transfer function without introducing input filter) by using Middlebrook's extra element theorem. According to the theorem, the input filter is considered as an extra element having impedance $Z_o(s)$ and the

modified $G_{vd}(s)$, may be written as in eqn. (2). The converter system with the addition of an input filter is illustrated as a block diagram in figure. 7-3. The above equation

implies that addition of an input filter modifies $G_{vd}(s)$ by a factor of $\frac{\left(1+\frac{Z_{0}(s)}{Z_{N}(s)}\right)}{\left(1+\frac{Z_{0}(s)}{Z_{D}(s)}\right)}$. Here,

 $Z_N(s)$ and $Z_D(s)$ are numerator and denominator of the input impedance, $Z_i(s)$ as seen from the converter with the addition of the input L-C filter comprising of L_i , C_{i1} and C_{i2} .



Figure 7.3: Generalized block schematic of a converter with an input filter in addition to

the output filter

$$G_{vd}^{\prime}(s) = \left(G_{vd}(s)\right) \frac{\left(1 + \frac{Z_{0}(s)}{Z_{N}(s)}\right)}{\left(1 + \frac{Z_{0}(s)}{Z_{D}(s)}\right)}$$
(7.28)

$$\|Z_o(s)\| \ll \|Z_N(s)\| \tag{7.29}$$

$$\|Z_o(s)\| \ll \|Z_D(s)\| \tag{7.30}$$

$$G_{\nu d}(s) = \frac{9.35*10^{-05}s + 3400}{4.1*10^{-07}s^2 + 0.001439s + 2}$$
(7.31)

$$Z_{o}(s) = \frac{1.62*10^{-25} s^{5} + 1.62*10^{-18} s^{4} + 1.782*10^{-16} s^{3}}{\frac{+ 1.674*10^{-15} s^{2} + 5.46*10^{-15} s^{1} + 6*10^{-15}}{8.1*10^{-23} s^{5} + 1.08*10^{-20} s^{4} + 1.62*10^{-14} s^{3}}}_{+ 1.62*10^{-13} s^{2} + 5.4*10^{-13} s^{1} + 6*10^{-13}}}$$
(7.32)

$$Z_N(s) = 1003 (7.33)$$

$$Z_D(s) = \frac{5.843*10^{-11}s^3 + 2.05*10^{-07}s^2 + 0.000285s}{2.85*10^{-09}s^2 + 1*10^{-05}s}$$
(7.34)

It is also stated in [30], [31], [39], that even with addition of input filter, converter would retain its original dynamics if inequalities (7.29) and (7.30) are satisfied. It implies, when the input filter is designed such that considerations (7.29) and (7.30) are met, the introduction of state variables associated with the input filter in dynamic equations would have no effect in the behavior of the dynamic equations obtained in Table 2. Thus, it may be seen from eqns. (7.28), (7.29) and (7.30) that if the inequalities in eqn. (7.29) and (7.30) are strictly followed while designing the input filter, the new converter duty ratio to output voltage transfer function, $G'_{vd}(s)$, would be same as original transfer function,

$$G_{vd}(s)$$
, since the factor of $\frac{\left(1+\frac{Z_O(s)}{Z_N(s)}\right)}{\left(1+\frac{Z_O(s)}{Z_D(s)}\right)}$ that is introduced due to addition of input filter [38]

would approximately become unity. Hence, it may be concluded from above discussions that these inequalities (7.29) and (7.30) serve as necessary design considerations for selection of parameters of input filter of center-point-clamped ac-ac buck converter. The values of designed input filter parameters along with the other converter parameters are tabulated in Table 3 in the following chapter. The converter transfer functions derived with the parameters in Table 3 are shown in above equations (7.30)-(7.33). The Bode plots of transfer functions, $Z_o(s)$, $Z_N(s)$, and $Z_D(s)$ are shown in figure 7.4. It may be seen from the Bode magnitude plots in figure 7.4 that the input filter parameters in Table 3 are designed such that the inequalities are satisfied. This implies from our argument that the Bode magnitude and phase plots for $G_{vd}(s)$ and $G'_{vd}(s)$ should be identical.



Figure 7.4: Bode plots for $Z_o(s)$, $Z_N(s)$ and $Z_D(s)$



Figure 7.5: Bode plot for $G_{vd}(s)$



Figure 7.6: Bode plot of uncompensated transfer function, $G'_{vd}(s)$



Figure 7.7: Bode plot of compensated transfer function $G'_{vd}(s) * G_c(s)$

The Bode plots of uncompensated transfer functions, $G_{vd}(s)$ and $G_{vd}'(s)$ are shown in figure 7.5 and figure 7.6 respectively. It may be seen from figures 7.5 and 7.6 that the Bode plots are almost identical. As may be seen from Bode plot in figure 7.7, the converter system is stable even after the introduction of an input filter. These Bode plots further verify the efficacy of the extra element theorem based methodology in determining the converter transfer function after introduction of an input filter. These plots also verify the significance of the necessary inequalities, (7.29) and (7.30) that are to be considered for the design of the input filter for Center-Point-Clamped AC-AC Converter in order to preserve the performance and stability of the converter system.

It has been verified that the Bode plot of the converter transfer function derived in Section 7.4 is same as the Bode plot of the converter transfer function obtained in this section through the above discussed Extra Element theorem. This further validates the function when an input side filter or an extra element is added to a converter.

CHAPTER 8 : SIMULATION RESULTS

8.1. Introduction

In this chapter, simulation results' verifying the operation and efficacy of the proposed Center-Point-Clamped AC-AC Direct Power Converter has been presented. Single phase $(2.4kV_{line-neutral})$ of the proposed converter for $4.16kV_{line-line}$, 600 kW application has been simulated using Matlab/Simulink. The IGBT switches have been considered to be ideal in the Simulink model. A simplified circuit schematic of the converter has been shown in figure.8-1.



Figure 8.1: Simplified circuit schematic of center-point-clamped ac-ac buck converter with input filter.

It may be seen from figure 8.1 that the input side inductor has been included in the simulation model. The design parameters for the simulation model have been discussed in the following sections.

8.2. Matlab-Simulink Model of Center-Point-Clamped AC-AC Converter

The matlab-simulink model of the proposed converter has been designed using the simulation parameters tabulated in Table 3.

V _{line-line}	4160 V rms
Р _{3-ф}	600 kW
v_{in} (line-neutral)	2400 V rms
Li	0.1e-3 H
R _{Li}	10e-3 Ω
<i>R</i> _{<i>i</i>1} , <i>R</i> _{<i>i</i>2}	3e+3 Ω
<i>C</i> _{<i>i</i>1} , <i>C</i> _{<i>i</i>2}	100e-6 F
R_{Ci1} , R_{Ci2}	1e-3 Ω
L _O	20.5е-3 н
R _{LO}	2.2e-3 Ω
Co	300e-6 F
R _{Co}	2.75e-3 Ω
$f_{ m sw}$	1.2kHz

Table 3. Simulation Parameters

The design of the simulation parameters have been analyzed in Chapter 6 and Chapter 7. The switching sequence for the proposed converter has already been discussed in Section 3.3. It may be seen from the simulation results that the switching sequence proposed for the center-point-clamped ac-ac buck converter implements input side capacitor voltage balancing as discussed in Section 3.4. A screenshot of the Matlab-Simulink model of the center-point-clamped ac-ac buck converter is shown in figure 8.2.



Figure 8.2: Matlab-Simulink model of center-point-clamped ac-ac buck converter with input filter.

The modulation strategy of the converter may be seen from Table 1. The operational simulation results have been shown for $v_{oref} = 0.3v_{in}$ and $v_{oref} = 0.8v_{in}$. As may be observed from figure 8.3, when desired output peak voltage is less than $0.5v_{in}$, the converter switches between 0 and $v_{in}/2$ sinusoidal waveform. Alternatively, it may be seen in figure 8.4 that, when desired output peak voltage is more than $0.5v_{in}$, the converter switches between $v_{in}/2$ and v_{in} sinusoidal waveforms. This method of operation offers superior spectral performance compared to conventional ac-ac buck converter which switches between 0 and v_{in} .



Figure 8.3: Unfiltered Output Voltage waveforms when $v_{oref} = 0.3 v_{in}$



Figure 8.4: Unfiltered Output Voltage waveforms when $v_{oref} = 0.8v_{in}$



Figure 8.5: Filtered Output Voltage waveforms when $v_{oref} = 0.8v_{in}$



Figure 8.6: Filtered Output Voltage waveforms when $v_{oref} = 0.8v_{in}$

It may be observed from figures 8.5 and 8.6 that the low pass output filter designed for the converter in Chapter 5 has removed the high frequency switching harmonics that may be seen in figures 8.3 and 8.4. Furthermore, it may also be seen from figures 8.7 and 8.8 that under all conditions, voltages across all switches are clamped to $v_{in}/2$ with the proposed topology. This verifies the implementation of the concept of input side capacitor voltage balancing as discussed in Section 3.4. Thus, it may be said that switches rated at half the input voltage can be used to implement the center-point-clamped ac-ac buck converter, thereby increasing the efficacy of the converter as compared to other prevalent ac-ac converter topologies which require switches rated at the input voltage.



Figure 8.7: Voltages across S2 and S3



Figure 8.8: Voltages across S_1 and S_4

The open loop simulation results comprise of the simulation results obtained without the implementation of the feedback compensator. These simulation results have been analyzed in this section. It may be seen from figures 8.9 and 8.10 that the filtered output voltage and current have negligible higher switching harmonics.



Figure 8.9: Filtered output voltage waveform for D=0.25



Figure 8.10: Filtered output current waveform for D=0.25



Figure 8.11: Input source voltage waveform for D=0.25



Figure 8.12: Waveform of input current drawn from the source for D=0.25
It may be seen from figures 8.11 and 8.12 that the designed input filter has removed the higher frequency current harmonics from being introduced into the unregulated power supply.



Figure 8.13: Filtered output voltage waveform for D=0.75



Figure 8.14: Filtered output current waveform for D=0.75

Simulation results shown in figures 8.13 and 8.14 verify the efficacy of the designed output filter in Region 2 (v_{in} to $v_{in}/2$) of converter operation.



Figure 8.15: Input source voltage waveform for D=0.75





The command following performance of the feedback compensator designed for the center-point-clamped ac-ac buck converter may be analyzed through figures 8.17 and 8.18.







Figure 8.18: Filtered output voltage waveform for $v_{oref} = 3000$ V

To illustrate the dynamics associated with feedback control, first change of operating point is simulated as follows:

In Region 1, initially the operating point is set at $v_{oref} = 1000$ V. After system reaches 1000V, the operating point is moved within same Region 1 to 1500V at 0.2 sec. This is shown in figure 8.19.



Figure 8.19: Change of operating point within Region 1 (1000V to 1500V)

In Region 2, initially the operating point is set at $v_{oref} = 2500$ V. After system reaches 2500V, the operating point is moved within same Region 2 to 3000V at 0.2 sec. This is shown in figure 8.20.



Figure 8.20: Change of operating point within Region 2 (2500V to 3000V) Finally, in Region 1, initially operating point is set at v_{oref} = 1500V. After system reaches 1500V, operating point is moved to Region 2 to 3000V at 0.2 sec. This is shown in figure 8.21.



Figure 8.21: Change of operating point from Region 1 (1500V) to Region 2 (3000V)

As may be seen from figures 8.19, 8.20, and 8.21, upon change of reference, the output voltage exhibits a small overshoot and then settles to the new value in two-three fundamental cycles. This also illustrates the efficacy of the designed feedback compensator in following the command reference input.

To evaluate the line regulation performance of center-point-clamped ac-ac direct converter under varying source voltages, input voltage is varied by +10% and -10% of 3400V. First, this line voltage is increased by 10% at 0.2 sec, and its effect on the output voltage waveform is shown in figure 8.22.



Figure 8.22: Line regulation: performance during 10% increase in line voltage

Next, the line voltage is decreased by 10% at 0.2 sec, and its effect on the output voltage waveform is shown in figure 8.22.



Figure 8.23: Line regulation: performance during 10% decrease in line voltage

As may be seen from figure 8.22, when the line voltage experiences a spike of 10%, the output voltage exhibits a minimal change. Similarly, as illustrated in figure 8.23, when the line voltage experiences sag of 10%, the output voltage exhibits a negligible effect. This illustrates the command following performance of the system.

Finally, to evaluate the load regulation performance of center-point-clamped ac-ac converter under varying load currents, output current is varied by +50% and -50% of 35A. First, this load voltage is decreased from full load by 50%, and its effect on the output voltage waveform is shown in figure 8.24.



Figure 8.24: Load regulation: performance during 50% decrease in load current.

Next, load current is increased from 50% to full load, and its effect on the output voltage waveform is shown in figure 8.25.



Figure 8.25: Load regulation: performance during 50% decrease in load current.

As may be seen from figure 8.24, when the load drops 50% at 0.2 sec, the output voltage exhibits a small overshoot and then decreases to the original value in two-three fundamental cycles. Similarly, when the load suddenly increases by 50% at 0.2 sec, the output voltage exhibits a small undershoot and then increases to the original value in two-three fundamental cycles. This is illustrated in figure 8.25. Thus, it may be observed that the center-point-clamped ac-ac buck converter reduces the voltage stress across the bi-directional switches, thereby facilitating the use of low rated power semiconductor devices for higher rated voltages. Also, it may be seen that the designed input filter eliminates the injection of higher order frequency harmonic current into the input side. Furthermore, it also verifies the efficacy of the command following performance of the designed feedback compensator.

CHAPTER 9 : EXPERIMENTAL RESULTS

9.1. Introduction

An experimental prototype has been built in the laboratory to illustrate the feasibility of the converter and verify the efficacy of feedback controller for the proposed Center-Point-Clamped AC-AC Buck Converter. Single phase $(120V_{line-neutral})$ of the proposed center-point-clamped ac-ac buck converter rated for $208V_{line-line}$, 10 kW applications has been constructed and tested. Experimental parameters are displayed in Table 4. The experimental set-up has been analyzed in the following section.

V _{line-line}	208 V rms
P _{3-φ}	10 kW
v_{in} (line-neutral)	120 V rms
L _i	0.1e-3
R _{Li}	10e-3 Ω
R_{i1} , R_{i2}	3e+3 Ω
C _{i1} , C _{i2}	100e-6 F
R_{Ci1} , R_{Ci2}	1e-3 Ω
L _o	197e-6 H
R _{LO}	106e-3 Ω
Co	3.5e-6 F
R _{Co}	2.75e-3 Ω
$f_{ m sw}$	20 kHz



9.2. Experimental Set-up Description

Figure 9.1: Picture of experimental prototype in the laboratory

A picture of the experimental prototype is shown in figure 9.1. The experimental set-up implements the center-point-clamped ac-ac buck converter using four bi-directional switches. Each bidirectional switch has been built using two IGBTs connected in anti-parallel fashion. The IGBT used in this set-up is IRGP4072DPbF, which is rated at 300V and 40A. The IGBT based bi-directional switches are driven by an isolated IGBT gate driver from CONCEPT Technologies, 2SC0106T2A0. This gate driver has been mounted on an interfacing base board also supplied by CONCEPT Technologies, 2BB0108T2A0. The gate drive pulses are supplied to the gate driver by National Instruments General Purpose Inverter Controller (GPIC). An interfacing circuitry acts as a buffer between the National Instruments (NI) GPIC and the gate driver to protect the NI GPIC against

current surge during fault conditions. The NI GPIC is an FPGA based controller designed by National Instruments (NI) mainly for quick prototyping of power electronic applications. The gate driver logic is designed using NI LabVIEW FPGA which is installed in the computer, as may be seen from figure 9.1. The designed NI LabVIEW program, popularly known as a Virtual Instrument (VI) is then compiled for errors and then on successful compilation, it is downloaded into NI GPIC. The NI GPIC has a base clock frequency of 40 MHz, which serves as the upper limit for the driver logic signal frequency. The concepts of instruction level parallelism may be introduced into the VI to run the driver logic signals on higher order frequency of clock signals.



Figure 9.2: Picture of LabVIEW VI Block Diagram window implementing the gate driver logic

A picture of the LabVIEW VI generating the gate driver logic has been shown in figure 9.2. The modulation strategy described in Table 1 has been implemented in the

LabVIEW VI Block Diagram window shown in figure 9.2. As may be seen from figure 9.2, there is a Boolean stroke selection button which facilitates the change in the gate driver logic based on the operating point's location in the region (0 to $v_{in}/2$ or $v_{in}/2$ to v_{in}) of operation. It implies that when the desired output voltage lies between (0 to $v_{in}/2$), the LabVIEW VI generates the driver logic signals corresponding to the switching sequence for (0 to $v_{in}/2$) as shown in Table 1. Similarly, in case of desired output voltage between (0 to $v_{in}/2$), the LabVIEW VI generates the driver logic signals corresponding to the switching sequence for (0 to $v_{in}/2$), the LabVIEW VI generates the driver logic signals corresponding to the two regions of operation are shown in figure 9.3 and 9.4.



Figure 9.3: Driver Logic Signal waveform for switches, S_1 , S_2 , S_3 , and S_4 when $v_{oref}=0.3v_{in}$



Figure 9.4: Driver Logic Signal waveform for switches, S_1 , S_2 , S_3 , and S_4 when $v_{oref}=0.8v_{in}$



Figure 9.5: User Interface for entering reference output voltage, v_{oref} command input.

Figure 9.5 shows the capacitive touch screen based user interface from Reachtech Technologies. The User Interface has been designed using Query Mark-up Language (QML). The User Interface comprises of a horizontal slider which can be adjusted to the desired value using our fingertip. This slider input may be used to generate reference voltage values ranging from 0V to 120V rms. The slider input is sensed in a LabVIEW VI running in the computer. The User Interface data is transferred to the computer via serial communication protocols. This data is then taken as the reference output voltage, v_{oref} input by a LabVIEW VI, as shown in figure 9.6.



Figure 9.6: Picture of the LabVIEW VI sensing the reference output voltage, v_{oref} from the slider input

The output voltage, v_o is scaled down by a transformer having a transformation ratio of 10:1. The scaled down output voltage is then sensed by NI GPIC. The sensed output voltage is then sent to a Phase Locked Loop which has been implemented in a LabVIEW VI, as may be seen in figure 9.7.



Figure 9.7: Picture of the LabVIEW VI sensing output voltage, v_o using Phase Locked Loop

The output of the Phase Locked Loop is the amplitude of the output voltage signal, v_o . It is then multiplied by an appropriate gain and then compared with the reference output voltage, v_{oref} to determine the error. This error is then passed to the designed feedback compensator to achieve closed loop operation of the converter. A picture of the implementation of the feedback controller in the LabVIEW VI has been displayed in figure 9.8.



Figure 9.8: Picture of the LabVIEW VI implementing the feedback controller

The experiments results, which have been analyzed in the following sections, are generated using Tektronix MSO-3014 Oscilloscope.

9.3. Operational Experimental Results

The modulation strategy of the converter may be seen from Table 1. The operational experimental results have been shown for $v_{oref} = 0.3v_{in}$ and $v_{oref} = 0.8v_{in}$. As may be observed from figure 9.9, when desired output peak voltage is less than $0.5v_{in}$, the converter switches between 0 and $v_{in}/2$ sinusoidal waveform. Alternatively, it may be seen in figure 9.10 that, when desired output peak voltage is more than $0.5v_{in}$, the converter switches between $v_{in}/2$ and v_{in} sinusoidal waveforms. It may be seen from figures 9.9 and 9.10, that the output voltage waveforms have high frequency switching harmonics in them.



Figure 9.9: Unfiltered Output Voltage waveforms when $v_{oref} = 0.3v_{in}$



Figure 9.10: Unfiltered Output Voltage waveforms when $v_{oref} = 0.8v_{in}$



Figure 9.11: Filtered Output Voltage waveforms when $v_{oref} = 0.3v_{in}$



Figure 9.12: Filtered Output Voltage waveforms when $v_{oref} = 0.8v_{in}$

It may be observed from figures 9.11 and 9.12 that the high frequency switching harmonics seen in figures 9.9 and 9.10 have been attenuated by the introduction of low pass output filter. These waveforms verify the efficacy of the designed output filter. Furthermore, it may also be seen from figures 9.13 and 9.14 that under all conditions, voltages across all switches are clamped to $v_{in}/2$ with the proposed topology. This verifies the practical realization of the concept of Center-Point-Clamping of input source voltage using the technique of input side capacitor voltage balancing as discussed in Section 3.4. Thus, it may be stated that switches rated at half the input voltage can be used to implement the center-point-clamped ac-ac buck converter, thereby increasing the efficacy of the converter as compared to other prevalent ac-ac converter topologies which require switches rated at the input voltage.



Figure 9.13: Voltages across S2 and S3



Figure 9.14: Voltages across S_1 and S_4

9.4. Open Loop Experimental Results

The open loop experimental results comprise of the oscilloscope screenshots that have been obtained before the introduction of the feedback controller. These experimental results have been examined in this section. It may be seen from figures 8.9 and 8.10 that the filtered output voltage and current have negligible higher switching harmonics.



Figure 9.15: Filtered output voltage waveform for D=0.25



Figure 9.16: Filtered output current waveform for D=0.25



Figure 9.17: Input source voltage waveform for D=0.25



Figure 9.18: Waveform of input current drawn from the source for D=0.25

It may be seen from figures 9.17 and 9.18 that the designed input filter has removed the higher frequency current harmonics from being introduced into the unregulated power supply.



Figure 9.19: Filtered output voltage waveform for D=0.75



Figure 9.20: Filtered output current waveform for D=0.75

Experimental results shown in figures 9.19 and 9.20 verify the efficacy of the designed output filter in Region 2 (v_{in} to $v_{in}/2$) of converter operation.



Figure 9.21: Input source voltage waveform for D=0.75



Figure 9.22: Waveform of input current drawn from the source for D=0.75

Experimental results shown in figures 9.21 and 9.22 demonstrate the effectiveness of input filter in eliminating the unwanted higher order frequency harmonics from being fed into the unregulated power supply in case of Region 2 (v_{in} to $v_{in}/2$) of converter operation.

9.5. Closed Loop Experimental Results

The dynamics associated with the feedback control have been analyzed in this section through the waveforms obtained after introduction of the feedback controller. To illustrate the robustness of the feedback controller, the waveforms pertaining to change of operating point in different regions of converter operation has been discussed here. At first, the operating point is set at $v_{oref} = 20$ V and then suddenly the operating point is changed to 50V within Region 1 (0 to V_{in}/2). The response of the feedback controller to this sudden change in operating point may be seen in figure 9.23.



Figure 9.23: Change of operating point within Region 1 (20V to 50V).

In Region 2, initially the operating point is set at $v_{oref} = 70$ V. After system reaches 70V, the operating point is moved within same Region 2 to 90V. This is shown in figure 9.24.



Figure 9.24: Change of operating point within Region 2 (70V to 90V).

Finally, in Region 1, initially operating point is set at $v_{oref} = 20V$. After system reaches 20V, operating point is moved to Region 2 to 90V. This is shown in figure 9.25.



Figure 9.25: Change of operating point from Region 1 (20V) to Region 2 (90V).

It may be seen from figures 9.23, 9.24, and 9.25, that upon change of operating point, the output voltage exhibits a sub-cycle transient and then settles to the desired operating point in one-two fundamental cycles. This verifies the efficacy of the command following performance of the feedback compensator.



Figure 9.26: Line regulation: performance during step increase in line voltage when $v_{oref} = 80$ V.



Figure 9.27: Line regulation: performance during step decrease in line voltage when $v_{oref} = 80$ V.

To evaluate the line regulation performance of center-point-clamped ac-ac converter under varying source voltages, input voltage is varied by +10% and -10% of 120Vac. First, this ac voltage is increased by 10%, and its effect on the output voltage waveform is shown in figure 9.26. Next, the line voltage is decreased by 10%, and its effect on the output voltage waveform is shown in figure 9.27.

As may be seen from figure 9.26, when the line voltage experiences a spike of 10%, the output voltage exhibits a minimal change. Similarly, as illustrated in figure 9.27, when the line voltage experiences sag of 10%, the output voltage exhibits a negligible effect. This illustrates the command following performance of the system.

Finally, to evaluate the load regulation performance of center-point-clamped ac-ac buck converter under varying load currents, load current is decreased, and its effect on the output voltage waveform is shown in figure 9.28.



Figure 9.28: Load regulation: performance during step decrease in load current when $v_{oref} = 80$ V.

Next, load current is increased, and its effect on the output voltage waveform is shown in figure 9.29.



Figure 9.29: Load regulation: performance during step increase in load current when $v_{oref} = 80$ V.

As may be seen from figure 9.28, when the load drops, the output voltage exhibits a small overshoot and then decreases to the original value in two-three fundamental cycles. Similarly, when the load suddenly increases, the output voltage exhibits a small undershoot and then increases to the original value in two-three fundamental cycles. This is illustrated in figure 9.29.

CHAPTER 10 : FAMILY OF CENTER-POINT-CLAMPED AC-AC CONVERTERS

10.1. Introduction

This chapter discusses the family of direct ac-ac power converters, which has been aptly termed as family of Center-Point-Clamped Direct AC-AC Power Converters. This family of converters involves Center-Point-Clamped Direct AC-AC Buck Converter, Center-Point-Clamped Direct AC-AC Boost Converter, Center-Point-Clamped Direct AC-AC Buck-Boost Converter, and Center-Point-Clamped Direct AC-AC Ćuk Converter. The main objective behind this novel family of converters is the use of these converter topologies to improve the efficiency of power electronic based power quality improving devices. The efficiency of these power quality devices are improved on employment of these novel topologies as this family of converters employ semiconductor devices rated at half the rating of semiconductor devices are well as the switching sequence of these novel converters have been discussed in the following sections. Simulation results verifying the operation methodologies of these converters have also been presented in the following sections.



Figure 10.1: Simplified Circuit Schematic of a Center-Point-Clamped Direct AC-AC Buck Converter

Figure 10.1 shows the simplified circuit schematic of the Center-Point-Clamped Direct AC-AC Buck Converter that belongs to the novel family of Center-Point-Clamped Direct AC-AC Power Converters. This converter topology has been discussed in detail in this thesis. The switching sequence has been described again in Table 5.

Command		
v_{oref}	between 0 and	between $v_{in}/2$
(desired	$v_{in}/2$	and v_{in}
output)		
State 1	S ₂ S ₃ closed	S ₁ S ₄ closed
State 1	$S_1 S_4$ open	$S_2 S_3$ open
State 2	S ₂ S ₄ closed	S ₂ S ₄ closed
State 2	$S_1 S_3$ open	$S_1 S_3$ open
State 3	$S_2 S_3$ closed	S ₁ S ₄ closed
State 5	S ₁ S ₄ open	$S_2 S_3$ open
State 1	$S_1 S_3$ closed	$S_1 S_3$ closed
State 4	$S_2 S_4$ open	S ₂ S ₄ open

Table 5: Switching Pattern of the Center-Point-Clamped Direct AC-AC Buck Converter

The operating principle of the converter topology has been analyzed elaborately in this thesis. Simulation results as well as experimental results verifying the working principle of this converter topology has been presented in this thesis. The simulation results verifying the two regions of converter operation has been shown in figure 10.2 and figure 10.3.



Figure 10.2: Filtered output voltage waveform for *D*=0.25



Figure 10.3: Filtered output voltage waveform for D=0.75

The popular direct ac-ac buck converter topologies employ semiconductor devices rated at the source voltage but the center-point-clamped direct ac-ac buck converter employs semiconductor switches rated at half the magnitude of converter input voltage. Thus, it may be concluded that the employment of lower voltage rated power semiconductor devices increases the power efficiency of the center-point-clamped direct ac-ac buck converter topology as compared to other popular buck converter topologies. This verifies the efficacy of the novel topology of center-point-clamped ac-ac buck converter.





Figure 10.4: Simplified Circuit Schematic of a Center-Point-Clamped Direct AC-AC Boost Converter

Figure 10.4 shows the simplified circuit schematic of the Center-Point-Clamped Direct AC-AC Boost Converter that belongs to the novel family of Center-Point-Clamped Direct AC-AC Power Converters. The switching sequence of this novel converter topology has been shown in Table 6.

Command v_{oref} (desired output)	between 0 and Mv_{in}
State 1	$S_1 S_4$ closed $S_2 S_3$ open
State 2	$S_1 S_4$ open $S_2 S_3$ closed

Table 6: Switching Pattern of the Center-Point-Clamped Direct AC-AC Boost Converter

It may be seen in Table 6 that when the bidirectional switches S_1 and S_4 are closed and the bidirectional switches S_2 and S_3 are open, the converter output voltage, v_o is zero. But when the bidirectional switches S_1 and S_4 are open and the bidirectional switches S_2 and S_3 are closed, the converter output voltage, v_o is Mv_{in} , where M represents the voltage conversion ratio of this converter topology, which is as follows:

M = 1/(1-D), where D is the duty ratio of this converter topology.

Here, $D = t_{on}/(t_{on}+t_{off})$, where t_{on} is the time duration during which the bidirectional switch is closed and t_{off} is the time duration during which the bidirectional switch is open.

The main objective behind this novel topology of Center-Point-Clamped Direct AC-AC Boost Converter is to reduce the voltage stress across the bidirectional switches of the converter. It may be observed that conventional AC-AC Boost Converter topologies require semiconductor devices rated at desired output voltage, i.e. v_o . But this novel converter topology employs semiconductor devices rated at half the voltage ratings as that of the conventional converter topologies, $v_o/2$.


Figure 10.5: Filtered output voltage waveform for *D*=0.25



Figure 10.6: Filtered output voltage waveform for D=0.75

Simulation results verifying the operation of the converter for duty ratio of D=0.25 and D=0.75 are presented in figures 10.5 and 10.6. These simulation results prove the efficacy of the Center-Point-Clamped Direct AC-AC Boost Converter.



10.4. Center-Point-Clamped Direct AC-AC Buck-Boost Converter

Figure 10.7: Simplified Circuit Schematic of a Center-Point-Clamped Direct AC-AC Buck-Boost Converter

Figure 10.7 shows the simplified circuit schematic of the Center-Point-Clamped Direct AC-AC Buck-Boost Converter that belongs to the family of Center-Point-Clamped Direct AC-AC Power Converters. The switching sequence of this novel converter topology has been shown in Table 7.

Command v_{oref} (desired output)	between 0 and Mv_{in}
State 1	$S_1 S_4$ closed $S_2 S_3$ open
State 2	$S_1 S_4$ open $S_2 S_3$ closed

Table 7: Switching Pattern of the Center-Point-Clamped Direct AC-AC Buck-Boost Converter

It may be seen in Table 7 that when the bidirectional switches S_1 and S_4 are closed and the bidirectional switches S_2 and S_3 are open, the converter output voltage, v_o is zero. But when the bidirectional switches S_1 and S_4 are open and the bidirectional switches S_2 and S_3 are closed, the converter output voltage, v_o is Mv_{in} , where M represents the voltage conversion ratio of this converter topology, which is as follows:

M = D/(1-D), where D is the duty ratio of this converter topology.

Here, $D = t_{on}/(t_{on}+t_{off})$, where t_{on} is the time duration during which the bidirectional switch is closed and t_{off} is the time duration during which the bidirectional switch is open.

The main objective behind this novel topology of Center-Point-Clamped Direct AC-AC Buck-Boost Converter is to reduce the voltage stress across the bidirectional switches of the converter. It may be observed that conventional AC-AC Buck-Boost Converter topologies require semiconductor devices rated at summation of the input and desired output voltages, i.e $v_0 + v_s$. But this novel converter topology employs semiconductor devices rated at half the voltage ratings as that of the conventional converter topologies, $(v_0 + v_s)/2$.



Figure 10.8: Filtered output voltage waveform for D=0.25



Figure 10.9: Filtered output voltage waveform for D=0.75

Simulation results verifying the operation of the converter for duty ratio of D=0.25 and D=0.75 are presented in figure 10.8 and 10.9. These simulation results verify the

operation methodology as well as efficacy of the Center-Point-Clamped Direct AC-AC Buck-Boost Converter.



10.5. Center-Point-Clamped Direct AC-AC Ćuk Converter

Figure 10.10: Simplified Circuit Schematic of a Center-Point-Clamped Direct AC-AC Ćuk Converter

Figure 10.10 shows the simplified circuit schematic of the Center-Point-Clamped Direct AC-AC Ćuk Converter that belongs to the novel family of Center-Point-Clamped Direct AC-AC Power Converters. The switching sequence of this novel converter topology has been shown in Table 8.

Command v_{oref} (desired output)	between 0 and Mv_{in}
State 1	$S_1 S_2 S_3 S_6 closed$ $S_4 S_5 open$
State 2	$\begin{array}{c} S_1 S_6 \text{ open} \\ S_2 S_3 S_4 S_5 \text{ closed} \end{array}$

Table 8: Switching Pattern of the Center-Point-Clamped Direct AC-AC Ćuk Converter

It may be seen in Table 8 that when the bidirectional switches S_1 , S_2 , S_3 and S_6 are closed and the bidirectional switches S_4 and S_5 are open, the capacitors, C_{o1} and C_{o2} transfer the energy to the load. But when the bidirectional switches S_1 and S_6 are open and the bidirectional switches S_2 , S_3 , S_4 and S_5 are closed, energy is transferred from the input side source to the capacitors, C_{o1} and C_{o2} . Here, *M* represents voltage conversion ratio of the converter topology, which is as follows:

M = D/(1-D), where D is the duty ratio of this converter topology.

Here, $D = t_{on}/(t_{on} + t_{off})$, where t_{on} is the time duration during which the bidirectional switch is closed and t_{off} is the time duration during which the bidirectional switch is open.

The main objective behind this novel topology of Center-Point-Clamped Direct AC-AC Ćuk Converter is to reduce the voltage stress across the bidirectional switches of the converter. It may be observed that conventional AC-AC Ćuk Converter topologies require semiconductor devices rated at summation of the input and desired output voltages, i.e. $(v_o + v_s)$. But this novel converter topology employs semiconductor devices rated at half the voltage ratings as that of the conventional converter topologies, $(v_o + v_s)/2$.



Figure 10.11: Filtered output voltage waveform for *D*=0.25



Figure 10.12: Filtered output voltage waveform for D=0.75

Simulation results verifying the operation of the converter for duty ratio of D=0.25 and D=0.75 are presented in figures 10.11 and 10.12. These simulation results prove the

working principle as well as efficacy of the Center-Point-Clamped Direct AC-AC Ćuk Converter.

CHAPTER 11 : CONCLUSIONS

Historically, PWM based direct ac-ac power converters have started to play an increasingly prominent role in Custom Power and Flexible AC Transmission Systems (FACTS) applications. In comparison with indirect ac-ac power converters, direct ac-ac power converters are smaller due to lack of bulky capacitors that are used in indirect ones to store energy in the intermediate stages. Also, they are cheaper due to the lack of intermediate stages which are present in indirect ones. These intermediate stages require more power semiconductor devices as well as reactive components. But the implementation of the prevalent direct ac-ac converters by utilities remains challenging due to lack of availability of power semiconductor devices at utility scale voltage and power ratings.

The novel topology of Center-Point-Clamped AC-AC Direct Power Converter offers a unique technique to employ power semiconductor devices rated at lower voltage and power ratings for direct ac-ac power converters rated at utility scale voltage and power ratings. This thesis analyzes afore mentioned novel topology of converter that offers a clamp to mid-voltage point at the source. It may be seen that proposed topology reduces voltage stress on power devices by half and improves the harmonic spectrum at the output. Simulation as well as experimental results verifying the operation of proposed configuration has been presented in this thesis. Overall system design including the power circuit design, dynamic analysis, and controller design has also been discussed here. An input filter has also been designed for the prevention of switching harmonics from being introduced into the input power source.

Dynamic analysis is performed using state space averaging and small signal perturbation techniques and system transfer function is derived. A compensator is designed to improve the dynamic characteristics of this converter and simulations are performed to evaluate the efficacy of this controller. It is shown that the converter reaches steady state within two-three fundamental cycles when reference voltage is changed instantaneously. Simulation results along with experimental waveforms illustrating operating point change, line and load regulation are also presented in the paper which further indicates the robustness of the designed closed loop controller.

Finally, a novel family of center-point-clamped ac-ac direct power converters has been introduced in this thesis. This family of power converters can be used to construct power quality devices at utility scale voltage and power ratings with low voltage and power rated semiconductor devices, thereby increasing the cost efficiency as well as power transfer efficiency of power quality devices.

REFERENCES

- [1] G. Venkataramanan, B. K. Johnson, and A. Sundaram, "An AC-AC power converter for custom power applications," *Power Delivery, IEEE Transactions on*, vol. 11, pp. 1666-1671, 1996
- [2] D. Divan, J. Sastry, A. Prasai, and H. Johal, "Thin AC converters-A new approach for making existing grid assets smart and controllable," in *Power Electronics Specialists Conference, 2008. PESC 2008. IEEE*, 2008, pp. 1695-1701.
- [3] A. Emanuel-Eigeles and J. Appelbaum, "AC Voltage Regulation by Means of Chopper Circuits," Electronic Letters, Vol. 4, Issue 2, 1968, pp. 26-28.
- [4] A. Mozdzer and B.K. Bose, "Three-phase AC Power Control using Power Transistors," IEEE Trans. on Industrial Applications, vol. IA-12, no. 5, 1976, pp. 499-505.
- [5] P. D. Ziogas, D. Vincenti, G. Joos, "A Practical PWM AC Controller Topology," IEEE-IAS'92 Conf. Rec., pp. 880-887.
- [6] S. Srinivasan and G. Venkataramanan, "Comparative Evaluation of PWM AC-AC Converters," IEEE-PESC'95 Conference Records, pp. 529-535.
- [7] B. Kwon, B. Min, and J. Kim, "Novel Topologies for AC Choppers," IEE Proceedings on Electric Power Applications, Vol. 143, Issue 4, July 1996, pp. 33-330.
- [8] G. Venkataramanan, "Three-Phase Vector Switching Converters for Power Flow Control," IEE Proceedings Electric Power Applications, Vol. 151, Issue 3, May 2004, pp 321-333.
- [9] M. Venturini, "A new sine wave in, sine wave out, conversion technique eliminates reactive elements," in Proceedings of Powercon 7, San Diego, CA, 1980, pp. E3-1-E3-15.
- [10] D. Divan and J. Sastry, "Voltage Synthesis Using Dual Virtual Quadrature Sources-A New Concept in AC Power Conversion," IEEE Trans. on Power Electronics, vol. 23, no. 6, 2008, pp. 3004-3013.
- [11] D. Divan, and J. Sastry, "Control of Multilevel Direct AC Converters," IEEE-ECCE'09 Conf. Proc., pp. 3077-3084.
- [12] T. Meynard, and H. Foch, "Multi-level Conversion: High Voltage Choppers and Voltage-Source Inverters," IEEE-PESC'92 Conference Records, pp. 397-403.

- [13] A. Nabae, I. Takahashi, and H. Akagi, "A New Neutral-Point-Clamped PWM Inverter," IEEE Trans. on Industrial Applications, vol. 17, no. 5, September/October 1981, pp.518-523.
- [14] R. D. Middlebrook and S. Ćuk, Advances in Switched-Mode Power Conversion. Pasadena: Teslaco, 1981, pp. 73–89.
- [15] C. Liu, B. Wu, N. R. Zargari, D. Xu, and J. Wang, "A novel three phase three-leg AC/AC converter using nine IGBTs," IEEE Trans. Power Electron., vol. 24, no. 5, pp. 1151–1160, May 2009.
- [16] T. Wijekoon, C. Klumpner, P. Zanchetta, and P. W. Wheeler, "Implementation of a hybrid ac-ac direct power converter with unity voltage transfer," IEEE Trans. Power Electron., vol. 23, no. 4, pp. 1918–1926, Jul.2008.
- [17] R. Lai, F. Wang, R. Burgos, Y. Pei, D. Boroyevich, B. Wang, T. A. Lipo, V. D. Immanuel, and K. J. Karimi, "A systematic topology evaluation methodology for high-density three-phase PWM ac-ac converters," IEEE Trans. Power Electron., vol. 23, no. 6, pp. 2665–2680, Nov. 2008.
- [18] N. A. Ahmed, K. Amei, and M. Sakui, "A new configuration of single phase symmetrical PWM ac chopper voltage controller," IEEE Trans.Ind. Electron., vol. 46, no. 5, pp. 942–952, Oct. 1999.
- [19] F. L. Luo and H. Ye, "Research on dc-modulated power factor correction ac/ac converters," in Proc. IEEE IECON, 2007, pp. 1478–1483.
- [20] X. P. Fang, Z. M. Qian, and F. Z. Peng, "Single-phase Z-source PWM ac-ac converters," IEEE Power Electron. Lett., vol. 3, no. 4, pp. 121–124, Dec. 2005.
- [21] Y. Tang, S. Xie, and C. Zhang, "Z-source ac-ac converters solving commutation problem," IEEE Trans. Power Electron., vol. 22, no. 6, pp. 2146–2154, Nov. 2007.
- [22] D. H. Jang and G. H. Choe, "Step-up/down ac voltage regulator using transformer with tap changer and PWM ac chopper," IEEE Trans. Ind. Electron., vol. 45, no. 6, pp. 905–911, Dec. 1998.
- [23] R. Stala, S. Pirog, A. Mondzik, M. Baszynski, A. Penczek, J. Czekonski, and S. Gasiorek, "Results of investigation of multicell converters with balancing circuit—Part II," IEEE Trans. Ind. Electron., vol. 56, no. 7, pp. 2620– 2628, Jul. 2009.
- [24] R. H. Wilkinson, T. A. Meynard, and H. T. Mouton, "Natural balance of multicell converters: The two-cell case," IEEE Trans. Power Electron., vol. 21, no. 6, pp. 1649–1657, Nov. 2006.

- [25] D. Chen and J. Liu, "The uni-polarity phase-shifted controlled voltage mode AC-AC converters with high frequency AC link," IEEE Trans. Power Electron., vol. 21, no. 4, pp. 899–905, Jul. 2006.
- [26] L. G. de Vicuna, M. Castilla, J. Miret, J. Matas, and J. M. Guerrero, "Sliding mode control for a single-phase ac/ac quantum resonant converter," IEEE Trans. Ind. Electron., vol. 56, no. 9, pp. 3496–3504, Sep. 2009.
- [27] D. M. Lee, T. G. Habetler, R. G. Harley, T. L. Keister, and J. R. Rostron, "A voltage sag supporter utilizing a PWM-switched autotransformer," IEEE Trans. Power Electron., vol. 22, no. 2, pp. 626–635, Mar. 2007.
- [28] IEEE Recommended Practices and Requirements for Harmonic Control in Electrical Power Systems. IEEE Standard 519-1992.
- [29] M.M. Swamy and S. Rossiter, "Case studies on mitigating harmonics in ASD IAS Annual Meeting", 1994: 685-692.
- [30] D.M. Mitchell, "Power line filter design considerations for dc-dc converters," IEEE Industry Application Magazine, Nov/Dec 1999, Vol.5, No.6, pp. 16-26.
- [31] R.D. Middlebrook, "Input Filter Considerations in Design and Application of Switching Regulators," IEEE Industry Applications Society Annual Meeting, 1976 Record, pp.366-382.
- [32] N. Sokal, "System Oscillations Caused by Negative Input Resistance at the Power Input Port of a Switching Mode Regulator, Amplifier, Dc/Dc Converter, or DC/AC Inverter," IEEE Power Electronics Specialists Conference, 1973 Record, pp. 138-140.
- [33] V. Vlatkovic, D. Borojevic and F. Lee, "Input Filter Design for Power Factor Correction Circuits," IEEE Transactions on Power Electronics, Vol 11, No.1, pp. 199-205, January 1996.
- [34] A. Kislovski, R. Redl, and N. Sokal, Dynamic Analysis of Switching-Mode DC/DC Converters, New York: Van Nostrand Reinhold, Chapter 10, 1991.
- [35] H. She, H. Lin, X. Wang, and L. Yue, "Damped input filter design of matrix converter," in Power Electronics and Drive Systems, 2009. PEDS 2009. International Conference on, Nov. 2009, pp. 672 –677.
- [36] Y. Levron, H. Kim, and R. W. Erickson, "Systematic Input Filter Design of Matrix Converter by Analytical Estimation of RMS Current Ripple", IEEE transactions on Power Electronics, vol. 29,no. 7, July 2014.

- [37] P.K. Bhowmik, S. Yellapragada, M. Manjrekar, "Dynamic Analysis and Controller Design for a Center-Point-Clamped AC-AC Converter", to be presented at Applied Power Electronics Conference and Exposition, APEC-2015, March 2015.
- [38] R.D. Middlebrook, "The N Extra Element Theorem", IEEE Transactions on Circuits and Systems-I: Fundamental Theory And Applications, Vol.45, No.9, September 1998.
- [39] R.D. Middlebrook, "Design Techniques for Preventing Input Filter Oscillations in Switched–Mode Regulators," Proceedings of Powercon 5, pp. A3.1-A3.16, May 1978.